

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Varthalitis, Petros; Sakkas, Stelios

Working Paper Public debt consolidation and its distributional effects

ESRI Working Paper, No. 629

Provided in Cooperation with: The Economic and Social Research Institute (ESRI), Dublin

Suggested Citation: Varthalitis, Petros; Sakkas, Stelios (2019) : Public debt consolidation and its distributional effects, ESRI Working Paper, No. 629, The Economic and Social Research Institute (ESRI), Dublin

This Version is available at: https://hdl.handle.net/10419/228277

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Working Paper No. 629

July 2019

Public Debt Consolidation and its Distributional Effects

Petros Varthalitis *ab and Stelios Sakkas c

Abstract: We build a dynamic general equilibrium model with heterogeneous households, namely Rich and Poor, and capitalskill complementarity structure in the production function, to study aggregate and distributional implications of fiscal consolidation policies when the government uses a rich set of spending and tax instruments. Fiscal policy is conducted through constrained optimized fiscal rules. Our results show that, in the long run, fiscal consolidation enhances both aggregate efficiency and equity; however, it may hurt Rich households depending on which fiscal instrument takes advantage of the fiscal space created. Along the transition, wage inequality significantly increases due to the capital skill complementarity structure of the production function. Specifically, this happens because debt consolidation crowds in capital and this favours Rich (skilled) households. On the other hand, the reduction in interest rates and government bonds lead to a decrease in Rich households income coming from capital and government bonds which eventually decrease income inequality. Finally, a rather novel finding is that the combination of asset and skill heterogeneity amplifies the increase in wage inequality in the early phase of fiscal consolidation.

*Corresponding Author: petros.varthalitis@esri.ie

Keywords: Debt consolidation, distributional effects, fiscal policy, optimized fiscal rules.

JEL Codes: E62, H52, H54, H63

Acknowledgements: We would like to especially thank Apostolis Philippopoulos for his guidance and helpful suggestions. We would also like to thank Judith Delaney, George Economides, Alejandro Garcia-Cintado, Nektaria Glynia, Ilias Kostarakos, Spyros Lazarakis, Duc Khuong Nguyen, Lefteris Roubanis, Rachel Slaymaker, Jesus Lopez Rodriguez, Simone Salotti and Vanghelis Vassilatos for many discussions and comments. We have also benefited from comments by seminar participants at the ASSET 2018 conference, Florence, Italy, the 9th GBCR 2018 conference, Manchester, UK, the 4th AMEF 2018, Thessaloniki, Greece, the Universidad Pablo de Olavide Macroeconomics Workshop, the ESRI Quarterly Macro Meet Up seminar series and the 23rd Annual ICMAIF 2019 conference, Rethymno, Greece. Any errors are ours. The views expressed are purely those of the authors and may not in any circumstances be regarded as stating an official position of the European Commission or ESRI.

a The Economic and Social Research Institute (ESRI), Dublin, Ireland

b Department of Economics, Trinity College Dublin, Ireland

c European Commission, Joint Research Centre (JRC), Seville, Spain

ESRI working papers represent un-refereed work-in-progress by researchers who are solely responsible for the content and any views expressed therein. Any comments on these papers will be welcome and should be sent to the author(s) by email. Papers may be downloaded for personal use only.

Public Debt Consolidation and its Distributional Effects^{*}

Stelios Sakkas[†]

(European Commission, Joint Research Centre (JRC), Seville)

Petros Varthalitis[§]

(Economic Social Research Institute (ESRI) and Trinity College Dublin)

June 26, 2019

Abstract

We build a dynamic general equilibrium model with heterogeneous households, namely Rich and Poor, and capital-skill complementarity structure in the production function, to study aggregate and distributional implications of fiscal consolidation policies when the government uses a rich set of spending and tax instruments. Fiscal policy is conducted through constrained optimized fiscal rules. Our results show that, in the long run, fiscal consolidation enhances both aggregate efficiency and equity; however, it may hurt Rich households depending on which fiscal instrument takes advantage of the fiscal space created. Along the transition, wage inequality significantly increases due to the capital skill complementarity structure of the production function. Specifically, this happens because debt consolidation crowds in capital and this favours Rich (skilled) households. On the other hand, the reduction in interest rates and government bonds lead to a decrease in Rich households' income coming from capital and government bonds which eventually decrease income inequality. Finally, a rather novel finding is that the combination of asset and skill heterogeneity amplifies the increase in wage inequality in the early phase of fiscal consolidation.

Keywords: Debt consolidation, distributional effects, fiscal policy, optimized fiscal rules JEL: E62, H52, H54, H63

^{*}Acknowledgements: We would like to especially thank Apostolis Philippopoulos for his guidance and helpful suggestions. We would also like to thank Judith Delaney, George Economides, Alejandro Garcia-Cintado, Nektaria Glynia, Ilias Kostarakos, Spyros Lazarakis, Duc Khuong Nguyen, Lefteris Roubanis, Rachel Slaymaker, Jesus Lopez Rodriguez, Simone Salotti and Vanghelis Vassilatos for many discussions and comments. We have also benefited from comments by seminar participants at the ASSET 2018 conference, Florence, Italy, the 9th GBCR 2018 conference, Manchester, UK, the 4th AMEF 2018, Thessaloniki, Greece, the Universidad Pablo de Olavide Macroeconomics Workshop, the ESRI Quarterly Macro Meet Up seminar series and the 23rd Annual ICMAIF 2019 conference, Rethymno, Greece. Any errors are ours. The views expressed are purely those of the authors and may not in any circumstances be regarded as stating an official position of the European Commission or ESRI.

[†]European Commission, Joint Research Centre (JRC), Seville; Calle Inca Garcilaso, 3, 41092 Sevilla, Spain, Email: stylianos.sakkas@ec.europa.eu.

[‡]The Economic and Social Research Institute (ESRI), Whitaker Square, Sir John Rogerson's Quay, Dublin 2, Ireland. Email: petros.varthalitis@esri.ie

[§]Department of Economics, Trinity College Dublin, Dublin 2, Ireland.

1 Introduction

The 2008 world financial crisis saw most European countries in a vulnerable fiscal position with high deficits and debts as a share of GDP, above the 3% and 60% of the Stability and Growth Pact limits.¹ As a result several European governments have been forced to take restrictive fiscal actions, the so called fiscal consolidation.

By now most studies find that fiscal consolidation entails an intertemporal trade-off for the main macroeconomic aggregates, see e.g. Coenen et al. (2008), Schwarzmüller and Wolters (2014), Philippopoulos et al. (2015) and (2017a). That is the early phase of fiscal consolidation requires cuts in spending and/or rises in taxes.² This is the short run fiscal pain. In the long run, once debt reduction has been achieved, the resulting fiscal space can be utilized to increase spending and/or reduce distortionary taxes. This is the long run fiscal gain. However, fiscal consolidation also has important distributional implications. For example, Furceri et al. (2015) find that during fiscal consolidation periods the lowest income and wealth quantiles of the population became worse off in terms of their disposable income.

The aim of this paper is to study the distributional and aggregate implications of fiscal consolidation policy. Since the aggregate effects have been thoroughly studied, our primary focus will be on the distributional effects. In particular, we examine the intertemporal effects of fiscal consolidation on the distribution of wages and income, whether this reform generates an equity efficiency trade off, and if yes, whether the magnitude of this trade off varies over different time horizons (i.e. short, medium and long run). Furthermore, we examine whether any distributional implications depend on the fiscal policy mix, on the source of heterogeneity incorporated in the model and the measure of inequality under consideration.

To study this, we develop a dynamic general equilibrium model with heterogeneous households calibrated to the Euro Area over the period 2001-2015. The model includes two types of households, capital-skill complementarity in the production function in the spirit of Krusell et al. (2000) and endogenous human capital accumulation. Households differ in the type of labor they supply and their access to capital and financial markets.³ In a standard model with representative Ricardian households, each agent can smooth out exogenous fiscal policy changes,

¹There is by now a tendency of declining public deficits in the Euro Area. This is reflected not only in statistical indicators but also in the number of countries that are still under the Excessive Deficit Procedure (EDP). The structural deficit in the EU was reduced markedly from 4.3 % to 1.0 % in the Euro Area. At the country level, in 2010 only five Member States recorded long run deficits below the 3% of GDP reference threshold, while, 22 did so in 2014. See European Commission (2017).

 $^{^{2}}$ In addition, consolidation efforts usually come at the expense of growth-friendly spending items such as spending on public investment and education which further harms the prospects of long term growth see European Central Bank (2017) for a discussion on the trade-offs between fiscal consolidation and reforms.

³There are different ways to introduce heterogeneous agents in DSGE models. For instance, Gali et al. (2007), Forni et al. (2009), Coenen et al. (2013), Schwarzmüller and Wolters (2017) use models in which a share of households does not have access to financial or/and capital markets (Ricardian vs Non-Ricardian). Households can also exhibit heterogeneity in terms of their impatience (patient vs impatient) as in Bilbiie et al. (2013), their labor market status (public vs private sector workers) as in Ardagna (2007) and Economides et al. (2012) or with respect to their education and skills as in Angelopoulos et al. (2014), Angelopoulos et al. (2017) and Gomes (2018).

like fiscal consolidation, through borrowing or lending. In addition, all households rely on the same sources of income, while, their labor and capital incomes are identical. Thus, there is no room for distributional analysis. To capture distributional implications, we depart from this framework and allow for non-uniform distributions of asset holdings and skills.

In particular, households that save in the form of government bonds, own physical capital and firms and supply skilled labor services are referred to as *Rich*. On the other hand, households that do not have access to financial and capital markets, i.e. they live hand-to-mouth, and supply unskilled labor services are referred to as *Poor*. In line with empirical evidence we assume that the asset and skill distributions are positively related. The heterogeneity in the skill supply of labor force gives rise to the so called skill premium which is the measure of wage inequality in our setup. The skill premium implies that *Rich* households receive relatively higher wages than *Poor* households. Finally, both households can invest time in education and accumulate human capital.

Regarding policy, government has a rather rich set of spending and tax instruments at its disposal. Particularly, government levies consumption, labor and capital taxes to finance productivity-enhancing spending like public investment and spending on education, utilityenhancing expenditures like government consumption, as well as transfer payments targeted to *Rich* and *Poor* households.

Our main policy experiment is fiscal consolidation. That is the economy starts from a steady state with high debt-to-GDP ratio, say 85.3%, which is the EA-19 data average over the fiscal stress period 2008-2014, and travels towards a new reformed steady state with lower debt-to-GDP ratio, say 60%. We experiment with various reformed economies varying the fiscal instrument which reaps the benefit of fiscal space after debt consolidation. Government can achieve the transition from the status quo to the new reformed steady state by implementing alternative fiscal consolidation policy mixes. Following most of the literature on debt consolidation we follow a rule-based approach to policy. In addition, we focus on constrained optimized fiscal policy rules so as our results do not depend on ad hoc differences in feedback policy coefficients. To do this, we follow Schmitt-Grohé and Uribe (2004, 2007) so that we limit our attention to a family of simple fiscal rules. This means that all the fiscal instruments can respond to the gap between public debt and the target of public debt as shares of output.

The structure of our model enables us to assess aggregate and distributional effects of fiscal consolidation in the long run and along the transition. To do this, we utilize the key endogenous variables of the model. For example, to quantify aggregate effects, we employ variables such as output and social welfare, while, to measure distributional effects, we find it natural to use variables like the net income of *Rich* and *Poor* households, inequality in net incomes and wages,⁴

⁴The reason that we employ these two different measures of inequality is the following. The vehicle of our analysis is the household budget constraint, which is the most straightforward way to understand how the different dimensions of inequality are linked with endogenous choices, e.g. labor supply and saving decisions, access to financial markets and fiscal policy changes. For a systematic and comprehensive analysis see Krueger et. al. (2010) and Heathcote et. al. (2010).

household specific welfare as well as relative welfare.

Our main results are as follows. In the long run, fiscal consolidation always enhances both efficiency and equity. Thus, once debt reduction has been achieved and fiscal space has been created, output increases, while, both wage and income inequality decrease. Looking at welfare, in contrast, fiscal consolidation is not always Pareto efficient. For example, such reform can be harmful for *Rich* households when government uses the post consolidation fiscal space to increase non productive government consumption or transfers targeted to *Poor* households. On the other hand, we find that fiscal consolidation is Pareto efficient, i.e. both *Rich* and *Poor* households get better off in the reformed steady state, when the government uses the resulting fiscal space to finance an increase in spending on public education and/or to finance a cut in distortionary taxation.

Along the transition, the government should cut non productive government consumption to reduce debt to its new lower target, while, keeping all the remaining fiscal instruments constant at their historical averages. This induces a temporary output contraction, however due to the crowding in effects, aggregate output quickly recovers above its pre consolidation level. Thus fiscal consolidation is costly on impact but efficient in the medium/long run.

On the other hand, the different driving forces of inequality in the model lead to interesting distributional implications that may differ across the various measures of inequality. In a nutshell, the skill premium increases, while, income inequality decreases. In terms of welfare, fiscal consolidation hurts only *Poor* households in the short run, while, it benefits *Rich* households.

More specifically, wage inequality significantly increases in the short/medium run due to the capital skill complementarity structure of the production function. The logic is the following, fiscal consolidation results in an increase in the physical capital stock and a reduction in relative skill supply which drive upward the wages of *Rich* households and increases the gap with the wage of *Poor* households in the short run.

However, income inequality decreases due to the asset heterogeneity between *Rich* and *Poor* households. Since fiscal consolidation induces a sharp reduction in government bonds and real rates, *Rich* households, who own assets and capital, see a decrease in their income coming from government bonds and capital. The latter leads to the deterioration in *Rich* households' total net income vis-à-vis *Poor* households' total net income despite the opposite movement in their wages and labor incomes.

A rather novel finding of this paper is that the combination of asset and skill heterogeneity amplifies the increase in wage inequality in the early phase of fiscal consolidation. *Poor* households rely only on labor income since they do not own assets and capital. Thus, in the transition phase of consolidation, they need to increase their labor supply to compensate the loss in their labor income due to the reduction in their wages. The increase in their hours worked pushes further downward *Poor* households' wages, and thus, widens the inequality with respect to *Rich* households' wages. Finally, in terms of welfare, fiscal consolidation is not Pareto efficient in the short run. Although it seems to enhance social welfare, this increase arises solely from welfare gains accruing to *Rich* households, while the welfare of *Poor* households decreases.

This paper contributes to the vast literature on fiscal consolidation by focusing mainly on its distributional implications. First, this paper is closely related to papers that study the aggregate and distributional implications of fiscal consolidation in dynamic general equilibrium models under ad hoc policy, e.g. Coenen et al. (2008), Economides et al. (2012), Almeida et al. (2013), Schwarzmüller and Wolters (2017) and Roubanis (2018).⁵ Second, our work is related to papers that compute optimal fiscal consolidation policies in DSGE models, e.g. Bi and Kumhof (2011), Cantore et al. (2012), Philippopoulos et al. (2015), (2017a) and (2017b) and Cardani et al. (2018). Third, our work is also related to papers that study other types of fiscal policy reforms in dynamic general equilibrium models with heterogeneous agents, like Garcia-Mila et al. (2010), Angelopoulos et al. (2013), Angelopoulos et al. (2014), Angelopoulos et al. (2017), Gomes (2018), Michaud and Rothert (2018), however, they do not focus on the distributional effects of fiscal consolidation.

In this paper, we employ a joint heterogeneity setup including asset and skill heterogeneity⁶ to assess fiscal consolidation. Moreover, we compute constrained optimized fiscal policy rules for a rich set of (utility- and productivity-enhancing) spending and tax instruments. Thus, we provide a systematic framework to assess the distributional implications, along with the aggregate effects, of fiscal consolidation policy.

The structure of the paper proceeds as follows. Section 2 presents the model and solves for the decentralized competitive equilibrium. Section 3 develops our calibration strategy and solves for the status quo steady state solution. Section 4 describes the main policy experiment, while, Section 5 presents our results. Section 6 analyzes the underlying transmission mechanism and Section 7 conducts sensitivity analysis. Section 8 concludes the paper. Algebraic details and additional results are in an Appendix.

2 The model

2.1 Informal description of the model

We develop a closed economy dynamic general equilibrium model which consists of households, firms and a government. The key feature of the model is household heterogeneity. Households differ in two aspects. First, in the type of labor they supply and second in their access to financial and capital markets. Thus we incorporate *ex ante* skill and wealth heterogeneity. In particular households that have access to capital and financial markets, supply skilled labor services and own private firms are referred to as *Rich*. Households that do not participate in capital and financial markets and supply unskilled labor services, thus, only consume their after-tax labor

⁵On the other hand, Forni et al. (2010), Cogan et al. (2013), Erceg and Lindé (2013), Pappa et al. (2015) and Economides et al. (2017) focus on the aggregate implications of debt consolidation under ad hoc policy.

⁶See e.g. He and Liu (2008) and Angelopoulos et al. (2014) and the references therein on the interaction among various fiscal policy reforms and distribution.

income referred to as Poor.⁷ In addition, both household types can accumulate human capital using a human capital production function à la Lucas while they derive utility from public consumption.

On the production side, firms use physical and public capital, skilled and unskilled labor in order to produce an homogeneous good. In the production sector we incorporate a nested CES-Cobb Douglas production function similar to Krusell et al. (2000) which exhibits capital-skill complementarity. As is known, this feature gives rise to the so called skill premium.

The government has a rather rich set of fiscal policy instruments at its disposal. In particular, it issues public debt and levies consumption, labor and capital income taxes to finance its stream of public expenditures, namely spending on public education and investment, government consumption and transfer payments which are allowed to be allocated unevenly between households. Finally, we assume that all agents have perfect foresight.

2.2 Population composition

The population size, N, is exogenous and constant. It is comprised by two types of households, i.e. Rich households indexed by the subscript $R = 1, 2, ..., N_R$ and Poor households indexed by the subscript $P = 1, 2, ..., N_P$ where $N_R > N_P$ and $N_P + N_R = N$ is the total size. No mobility or occupational change is possible between the two types. There are also $f = 1, 2, ..., N^f$ firms. For notational convenience, we assume also that $N^f = N_R$.⁸

2.3 *Rich* households

Each Rich household, R, maximizes its discounted lifetime utility given by:

$$V_{R,0} \equiv \sum_{t=0}^{\infty} \beta^{t} U_{R,t} \left(c_{R,t}, z_{R,t}, g_{t}^{c} \right)$$
(1)

where $c_{R,t}$ and $z_{R,t}$ are consumption and leisure of each household, g_t^c is per capita utilityenhancing government consumption⁹ and $0 < \beta < 1$ is the discount rate. The period utility function $U_{R,t}(.)$ is increasing and strictly concave in all its arguments. For our numerical solutions the period utility function is given by:

$$U_{R,t}\left(c_{R,t}, z_{R,t}, g_{t}^{c}\right) = \mu_{1} log\left(c_{R,t} + \xi g_{t}^{c}\right) + \mu_{2} log\left(1 - e_{R,t} - l_{R,t}\right)$$
(2)

where μ_1, μ_2 are preference parameters, ξ measures the degree of substitutability between private and public consumption, e.g. if $\xi > 0$ (< 0) private and public consumption are substitutes

⁷We follow Michaud and Rothert (2018) by referring to the two types of households as *Rich* and *Poor*.

⁸Following the related literature, e.g. Angelopoulos et al. (2014) and Michaud and Rothert (2018), we work with a discrete number for population and its segments. We report that our results do not depend on this assumption.

⁹Notice that $g_t^c \equiv \frac{G_t^c}{N}$ where G_t^c is aggregate utility-enhancing government consumption.

(complements). Households are endowed with a normalized time unit:

$$z_{R,t} + e_{R,t} + l_{R,t} = 1 \tag{3}$$

where $l_{R,t}$ and $e_{R,t}$ is time devoted to labor and education respectively. The within period budget constraint of each *Rich* household, *R*, is:

$$(1+\tau_t^c)c_{R,t} + i_{R,t} + d_{R,t} = (1-\tau_t^k)(r_t k_{R,t} + \pi_{R,t}) + (1-\tau_t^l)w_{R,t}l_{R,t}h_{R,t} + tr_{R,t} + r_t^b b_{R,t}$$
(4)

where $i_{R,t}$, is private investment in physical capital, $k_{R,t}$, the beginning of period physical capital, $b_{R,t}$, the beginning of period government bonds whose gross returns are r_t and r_t^b respectively, $h_{R,t}$, is the beginning of period human capital, $d_{R,t}$, is savings in the form of government bonds, $\pi_{R,t}$ is dividends received from private firms, $w_{R,t}$ is the wage rate earned by *Rich* households, $tr_{R,t} \equiv \frac{TR_{R,t}}{N_R}$ is public transfers to each *Rich* household and $0 < \tau_t^k, \tau_t^l, \tau_t^c < 1$, are tax rates on capital income, labor income and consumption respectively. *Rich* households supply skilled labor services while *Poor* households supply unskilled labor services to firms. Thus, *Rich* households receive a relatively higher wage $w_{R,t} > w_{P,t}$ than *Poor* households (for more details see section 2.6).¹⁰ To allow for productive education expenditures we use a human capital production function as in Lucas (1988) and Glomm and Ravikumar (1997).¹¹ Therefore, individual human capital is augmented by time spent in education, $e_{R,t}$, and by public spending on education, $g_{R,t}^e \equiv \omega g_t^e$, which is a fixed share ω of per capita public spending on education, g_t^e .¹² The law of motion of human capital of *Rich* household, *R*, is:

$$h_{R,t+1} = (1 - \delta^h) h_{R,t} + B_R \left[(e_{R,t})^\theta \left(g_{R,t}^e \right)^{1-\theta} \right]^{x_R}$$
(5)

where $B_R > 0$, $\theta \in (0, 1)$, $x_R < 1$ are productivity parameters, and δ^h is the depreciation rate of human capital. Following He and Liu (2008), $x_R < 1$, captures decreasing returns to scale in the production of new human capital. The laws of motion of physical capital and government bonds for each R are:

$$k_{R,t+1} = (1 - \delta^k) k_{R,t} + i_{R,t} \tag{6}$$

$$b_{R,t+1} = b_{R,t} + d_{R,t} \tag{7}$$

Each Rich household in any given period t, chooses $c_{R,t}$, $e_{R,t}$, $l_{R,t}$, $k_{R,t+1}$, $b_{R,t+1}$ and $h_{R,t+1}$ to maximize its lifetime utility subject to the constraints (4) (in which we incorporate (6) and

¹⁰Throughout the rest of the paper labor provided by *Rich* households referred to as skilled labor and labor provided by *Poor* households referred to as unskilled labor. As we explain in section 3.1 there is adequate empirical evidence associating wealth with skills.

¹¹Alternatively, Trabandt and Uhlig (2011) use a learning by doing specification by including hours of work and education as inputs. On the contrary, in our model, as in Daniel and Gao (2015), we allow for a production function that combines a time input and a good input so as to assess the effects of public education spending as an additional productivity enhancing fiscal instrument.

¹²Similarly, $g_t^e \equiv \frac{G_t^e}{N}$ where G_t^e is total public spending on education.

(7)) and (5) taking factor prices and policy as given. Defining as $\lambda_{R,t}$ and $\psi_{R,t}$ the Lagrange multipliers associated with (4) and (5) respectively. The first-order conditions are given in Appendix A.1.

2.4 *Poor* households

Each *Poor* household, *P*, maximizes its expected discounted lifetime utility given by:

$$V_{P,0} \equiv \sum_{t=0}^{\infty} \beta^{t} U_{P,t} \left(c_{P,t}, z_{P,t}, g_{t}^{c} \right)$$
(8)

The period utility function $U_{P,t}(.)$ is increasing and strictly concave in all its arguments where we use the same functional form for preferences and the same normalized time unit as above.¹³ The within period budget constraint of each *Poor* household is given by:

$$(1 + \tau_t^c) c_{P,t} = (1 - \tau_t^l) w_{P,t} l_{P,t} h_{P,t} + t r_{P,t}$$
(9)

where $w_{P,t}$ is the wage rate received by *Poor* households, $tr_{P,t} \equiv \frac{TR_{P,t}}{N_P}$ is public transfers to each *Poor* household. Contrary to *Rich* households, *Poor* households do not have access to financial and capital markets thus the only source of income is the labor income.¹⁴ In addition they supply unskilled labor, and thus receive lower wages than *Rich* households. As before, the law motion of human capital of each household of type, *P*, is:

$$h_{P,t+1} = (1 - \delta^h)h_{P,t} + B_P \left[(e_{P,t})^{\theta} \left(g_{P,t}^e \right)^{1-\theta} \right]^{x_P}$$
(10)

where $g_{P,t}^e \equiv (1-\omega) g_t^e$ is the amount of total public spending on education services enjoyed by each $P.^{15}$ Each *Poor* household, P, maximizes its lifetime utility in any given period t by choosing $c_{P,t}, e_{P,t}, l_{P,t}$, and $h_{P,t+1}$ subject to the constraints (9) and (10) taking factor prices and policy as given. Defining as $\lambda_{P,t}$ and $\psi_{P,t}$ the Lagrange multipliers associated with (9) and (10) respectively. The first-order conditions are given in Appendix A.2.

2.5 Firms

There are $f = 1, 2..., N^f$ identical firms owned by the *Rich* households. Each firm, f, acts competitively taking prices as given. Firm's objective is to maximize their profits:

$$\pi_t^f \equiv y_t^f - r_t k_t^f - w_{R,t} l_{R,t}^f - w_{P,t} l_{P,t}^f \tag{11}$$

¹³Notation and functional forms of *Poor* households are analogous to *Rich* households.

 $^{^{14}}$ In section 7.2 we relax this assumption.

¹⁵This is meant to be not only formal education (i.e. secondary or tertiary education spending), but could resemble other types of educational programmes such as vocational training, on-the-job learning, continuing professional development programmes among others. This type of investment is of special importance for the less skilled or less wealthy members in the society since it increases their productivity and labor earnings.

where y_t^f is firm f's output. Firms utilize four factors inputs to produce an homogeneous good, i.e. physical capital, k_t^f , skilled labor services rented from *Rich* households, $l_{R,t}^f$, unskilled labor services rented from *Poor* households, $l_{P,t}^f$, and per capita public capital, k_t^g . Production is given by the following constant returns to scale (CRS) and constant elasticity of substitution (CES) function similar to Krusell et al. (2000):

$$y_t^f = A\left[m\left(l_{P,t}^f\right)^\sigma + (1-m)\left(\rho\left(k_t^f\right)^v + (1-\rho)\left(l_{R,t}^f\right)^v\right)^{\frac{\sigma}{v}}\right]^{\frac{\sigma}{\sigma}} \left(k_t^g\right)^{1-\alpha}$$
(12)

where A > 0 is scale parameter, $0 < \alpha, \rho, m < 1$ are factor inputs share parameters and $\sigma, v \leq 1$ are parameters governing factor elasticities (see below for more details). Each firm f maximizes its profits (11) subject to its production function (12) by choosing k_t^f , $l_{R,t}^f$, $l_{P,t}^f$. First order conditions are given in Appendix A.3.

2.6 Skill premium

In our setup the skill premium is defined as the ratio of the wage rate earned by the *Rich* household over the wage rate earned by the *Poor* household. Combining the first order conditions with respect to $l_{R,t}^f$ and $l_{P,t}^f$ in the Appendix A.3 gives the following condition for the skill premium:

$$\frac{w_{R,t}}{w_{P,t}} = (1-\rho) \frac{1-m}{m} \left(\frac{l_{P,t}^f}{l_{R,t}^f}\right)^{1-\sigma} \left[\rho \left(\frac{k_t^f}{l_{R,t}^f}\right)^{\nu} + (1-\rho)\right]^{\frac{\sigma}{\nu}-1}$$
(13)

The different roles in the production function for skilled (*Rich*) and unskilled (*Poor*) labor give rise to the so called skill premium meaning that $\frac{w_R}{w_P} > 1$. In section 3.1 we calibrate the associated parameters in the production function so that the implied factor input elasticities and the resulting skill premium are in line with empirical studies. The elasticities of substitution between physical capital and skilled labor and between skilled and unskilled labor is $\frac{1}{1-\sigma}$ whereas the elasticity between capital and skilled labor is $\frac{1}{1-\nu}$. This formulation implies that as long as $\sigma > v$ the production function exhibits capital-skill complementarity. Moreover, this specification implies that the skill premium will be *ceteris paribus* increasing in physical capital to skilled labor ratio, $\frac{k_{R,t}^f}{l_{R,t}^f}$ (known as the capital-skill complementarity effect) and decreasing in the skilled to unskilled labor ratio, $\frac{l_{R,t}^f}{l_{P,t}^f}$ (known as the relative skill supply effect). For more details see in Lindquist (2004) and Dolado et al. (2018).¹⁶

¹⁶Notice that
$$\frac{\partial \left(\frac{w_R}{w_P}\right)}{\partial \left(\frac{k_R^f}{l_R^f}\right)} > 0$$
 as long as $\sigma > v$ and $0 < \rho$, $m < 1$ while $\frac{\partial \left(\frac{w_R}{w_P}\right)}{\partial \left(\frac{l_R^f}{l_R^f}\right)} < 0$ as long as $\sigma < 1$ and $0 < \rho$, $m < 1$.

2.7 Government

The within-period government budget constraint is given by (in aggregate terms):

$$G_t^c + G_t^i + G_t^e + TR_{R,t} + TR_{P,t} + (1 + r_t^b)B_t = B_{t+1} + T_t$$
(14)

where G_t^c is total public spending on utility-enhancing government consumption, G_t^i is total spending on public investment G_t^e is total public spending on education, $TR_{R,t}$ and $TR_{P,t}$ are respectively transfers to *Rich* and *Poor* households, $B_t \equiv N_R b_{R,t}$ denotes the beginning-of-period stock of government bonds, and T_t denotes total tax revenues defined as:

$$T_t \equiv \tau_t^c \left(N_R c_{R,t} + n_P c_{P,t} \right) + \tau_t^k N_R \left(r_t k_{R,t} + \pi_{R,t} \right) + \tau_t^l \left(N_P w_{P,t} l_{P,t} h_{P,t} + N_R w_{R,t} l_{R,t} h_{R,t} \right)$$
(15)

Each period t government sets eight fiscal instruments, i.e. five public spending instruments, namely utility-enhancing spending, public education, public investment and agent specific transfers to *Rich* and *Poor* households and three tax instruments, namely capital, labor and consumption taxes. We assume, unless otherwise stated, that the residual policy instrument is public debt. The law motion of aggregate public capital is given by:

$$K_{t+1}^g = (1 - \delta^g) K_t^g + G_t^i$$
(16)

For notational convenience, in what follows, we use per capita quantities denoted with small case letters for example, $k_t^g \equiv \frac{K_t^g}{N}$, while we define spending instruments in terms of their output shares, i.e. $s_t^{g^i} \equiv \frac{G_t^i}{N^f y_t^f} = \frac{Ng_t^i}{N^f y_t^f}$, $s_t^{g^e} \equiv \frac{G_t^e}{N^f y_t^f} = \frac{Ng_t^e}{N^f y_t^f}$, $s_t^{g^e} \equiv \frac{G_t^e}{N^f y_t^f} = \frac{Ng_t^e}{N^f y_t^f}$, $s_t^{tr_P} \equiv \frac{R_{P,t}}{N^f y_t^f} = \frac{N_R tr_{P,t}}{N^f y_t^f}$, $s_t^{tr_R} \equiv \frac{TR_{R,t}}{N^f y_t^f} = \frac{N_R tr_{R,t}}{N^f y_t^f}$ and total tax revenues in per capita terms, $\tau_t \equiv \frac{T_t}{N}$. Also, we express the number of Rich, Poor households and firms in terms of their total population share, i.e. $n_R \equiv \frac{N_R}{N}$, $n_P \equiv \frac{N_P}{N} = 1 - n_R$, $n^f \equiv \frac{N^f}{N} = n_R$ respectively.

2.8 Fiscal policy rules

Fiscal policy sets its spending-tax instruments following simple fiscal policy rules. In particular, we allow all the main policy instruments $\phi_t = \{s_t^{g^c}, s_t^{g^i}, s_t^{g^e}, s_t^{tr_P}, s_t^{tr_R}, \tau_t^c, \tau_t^l, \tau_t^k\}$ to react to the public debt to GDP ratio, $q_{t-1} \equiv \frac{n_R b_t}{n^J y_{t-1}}$, as deviation from a target according to a simple linear rule:

$$\phi_t - \phi^* = \gamma_q \left(q_{t-1} - q^* \right) \tag{17}$$

where ϕ^*, q^* denote fiscal policy targets and γ_q are feedback policy coefficients. If ϕ_t is a spending instrument then, $\gamma_q \leq 0$, and $\gamma_q \geq 0$ if ϕ_t is a tax instrument (see equations (60)-(67) in Appendix).

2.9 Decentralized Competitive Equilibrium (DCE)

DCE is defined as a sequence of allocations, prices and policies such that: (i) both household types maximize their lifetime welfare, (ii) firms maximize their profits, (iii) goods, capital, labor, dividends and bond markets clear, (iv) all constraints are satisfied, (v) policymakers follow fiscal rules in Section 2.8.

We end up with a non-linear dynamic equilibrium system summarized by 39 equations in 39 unknowns $\{y_t^f, c_{R,t}, c_{P,t}, i_{R,t}, d_{r,t}, k_{R,t+1}, h_{R,t+1}, h_{P,t+1}, b_{R,t+1}, e_{R,t}, e_{P,t}, l_{R,t}, l_{P,t}, r_t^b, \pi_{R,t}, \lambda_{R,t}, \lambda_{P,t}, \psi_{P,t}, k_t^g, r_t, w_{R,t}, w_{P,t}, q_t, g_t^c, g_t^i, g_t^e, tr_{R,t}, tr_{P,t}, g_{P,t}^e, g_t^{g^i}, s_t^{g^c}, s_t^{g^e}, s_t^{tr_P}, s_t^{tr_R}, \tau_t^c, \tau_t^l, \tau_t^k, \tau_t^k\}$. This is given the initial conditions for the state variables and the values of the feedback fiscal policy coefficients in the associated fiscal policy rules. We present the clearing market conditions and the full equilibrium system in the Appendices B and C respectively.¹⁷

2.10 Plan of the rest of the paper

In the rest of the paper we work as follows. First, using common structural parameter values and fiscal policy data from the EA-19 over the period 2001-2015 we solve for the steady state solution of the model in Section 3. We explain our calibration strategy in Section 3.1. The long run solution is computed in Section 3.2. Throughout the paper we refer to this solution as the status quo economy. In our policy experiments we use this solution as point of departure in order to evaluate debt consolidation policies under constrained optimized fiscal rules. Our fiscal policy experiment is described in Section 4.

Second, we compute various steady state reformed economies in which public debt to GDP ratio is lower. In the reformed economies thanks to public debt reduction one fiscal instrument can adjust in the reformed steady state to reap the benefit of the fiscal adjustment (fiscal gain). We study various reformed economies depending on which fiscal instrument adjusts in the new steady state. Aggregate and distributional long run effects are computed in Section 5.1.

Third, we compute the transitional dynamics from the status quo economy with high debt to GDP ratio to a reformed economy with lower public debt to GDP ratio. Aggregate and distributional effects in the transition are presented in Section 5.2.

In Section 6 we analyze the transmission mechanism of debt consolidation under the constrained optimized fiscal policy. Finally, we conduct a sensitivity analysis in Section 7.

¹⁷As the DCE system presented in the Appendix C shows, policy instruments distort first order conditions and resource constraints in several ways. These distortions are referred to as wedges in the related macroeconomic literature (see e.g. Chari et al., 2007). In particular, because of consumption taxes, labor taxes, capital taxes, government spending on infrastructure and government spending on education, there are wedges in agents' labor supply and investment decisions, while, because of government spending on infrastructure, education, utility-enhancing services and transfer payments, there are also wedges in budget constraints and resource constraints. This is typical of economic environments without lump-sum policy instruments. Here, however, the aim is to investigate the implications of debt consolidation by using the full model in hand, rather than to identify the role of each friction/wedge in a historical episode.

3 Calibration and status quo long run equilibrium

In this section we discuss our calibration choices and present the steady state solution of the model.

3.1 Parameter values and fiscal policy data

In Table 1a we report the values of the structural parameters. In Table 1b we report the fiscal policy instruments values using fiscal data averages for the EA-19 over the period 2001-2015. Parameter values and fiscal policy instruments are chosen so that the model's long run solution targets various observed key macroeconomic ratios of the EA-19 economy (see Table 2). We employ data from the following resources: AMECO database of the European Commission, Eurostat COFOG (Classification of Functions of Government), OECD Education at a Glance (EAG), LFS (labor Force Survey), EU-SILC (Social Income and Living Conditions), Household Finance and Consumption Survey (HFCS), SES (Structure of Earnings). Below we analyze in detail our calibration strategy.

Population shares In the present model households differ along two dimensions, i.e. access to financial/capital markets and skills. We set $n_P = 0.3$ and $n_R = 0.7$ so that 30% of total population does not participate in capital and financial markets which is within the range reported by Coenen et al. (2008). This is in line with data on household savings in HFCS which reports that the asymmetric savings distribution is also reflected in income distribution. For instance, in the Euro Area the richest 20% income group holds over 60% of total savings. Regarding the skills distribution, as is defined by the International Standard Classification of Education (ISCED), data from Eurostat indicate that in the EA-19 a range from 30% to 35% has at least attained lower secondary education (i.e. 10 years of education) while the rest has attained at least post-secondary non-tertiary and tertiary education which roughly matches our parameter choices for n_R and n_P . Finally, data from EU-SILC report that high income groups as well as high savings groups in the population exhibit relatively higher educational attainment rates. Thus, we deem there exists enough evidence to associate savings and income with skills and education.

Preferences and parameters common to all agents The time discount factor β is set so as to give an annual real interest rate equal to 2.25% which is consistent with data on EA-19 (see AMECO database). The preference parameters $\{\mu_1, \mu_2, \xi\}$ are calibrated so that the weighted average of skilled and unskilled hours worked is 0.232. It also implies that in steady state *Rich* households devote more time to labor relative to *Poor* (see Table 2 in the next subsection).¹⁸ We set the depreciation rates of physical and public capital $\{\delta^k, \delta^g\}$ equal to 6%, as in Coenen

¹⁸This is in accordance with the Eurostat LFS data which reports that the average hours worked per week of full time employment is 41.5 hours out 168 hours of the whole week. This gives an average labor time of 0.247 which is almost equal to our calculated value. The same survey reports that workers occupied in skilled professions (e.g. managers, professionals, engineers) record higher weekly hours worked than than those of unskilled professions (e.g. clerical stuff, technicians etc). Our numerical solution is consistent with these data.

et al. (2008). Given that there is not a clear consensus on the magnitude of the depreciation of human capital we assume $\delta^h = \delta^k = \delta^g = 6\%$.

Production We normalize the scale parameter A to 1. We calibrate the parameter, v = -0.553, which governs the elasticity of substitution between capital and skilled labor, $\frac{1}{1-v} = 0.6439$, and the parameter, $\sigma = 0.323$, which governs the elasticity of substitution between capital/skilled labor and unskilled labor, $\frac{1}{1-\sigma} = 1.4771$, jointly so as to match the target of 1.55 for the value of skill premium which is consistent with the data.¹⁹ Our calibrated values lie within the range of values estimated in Krusell et al. (2000) and Maliar et al. (2017). Then, we set the remaining parameters of the production function, $\{\rho, m\}$, so that the model's status quo solution is consistent with data on factor inputs shares such as labor income share, capital income share. The choice of the parameter α along with the depreciation rate of physical and public capital imply a physical capital to GDP ratio of 2.97 and a public capital to GDP ratio of 0.51.

Human capital Next, we set the parameters governing the production of new human capital of each household type. The sets of parameters $\{B_R, B_P\}$ and $\{x_R, x_P\}$ both relate to technology and ability in the creation of new human capital and skills. For this reason, similar to He and Liu (2008) and Angelopoulos, Benecchi and Malley (2017) we normalize B_R and B_P equal to 1 and let $x_R = 0.40 > x_P = 0.35$ to capture any differences in ability between the two household types. In essence, this choice reflects the standard Mincerian logic which assumes that *Rich* households due to their higher education status, obtain higher returns. We set the value of the elasticity parameter of education time with respect to new human capital θ equal to 0.8 so that $1 - \theta = 0.2$ as in Blankenau et al. (2004). This implies that households devote 13% on average of their time endowment to skill enhancing activities. Note that in the model both time spent on education and public education spending are meant to be post-formal schooling. This implies that both agents have already acquired a minimum of 10 years of education.²⁰

Policy To set the long run values of fiscal variables we employ data from Eurostat for EA-19 over 2001-2015. In particular, we set the long run values of tax instruments equal to the associated 2001-2015 average effective average tax rates, i.e. effective tax rate on consumption is equal to 19.8%, effective tax rate on labor is equal to 46.75% and effective tax rate on capital is equal to 36.8%.²¹ We set public debt to GDP ratio in the status quo economy equal to

¹⁹The value of the skill premium typically ranges from 1.45 to 1.80 depending on the country, timeframe and method of measurement (for an overview of inequality measures see Krueger et al. (2008) while for measurement issues see Crivellaro (2016) and OECD Education at a Glance reports (2008-2018)). For our target value we use OECD's Relative Earnings by educational attainment indicator.

 $^{^{20}}$ Data on the share of time that individuals spend on education are not available at this level of heterogeneity. Thus, we use as proxy the following formula. Skilled individuals devote on average at least 7 additional years to post-formal schooling educational activities. Under the assumption that the average retirement age is 63 years old and that working life begins at 16 years old, this implies 47 years of working life duration or an additional 14.8% of their time devoted to education (7/47 years). Similarly assuming that unskilled agents will devote 3 additional years to obtain at least an upper-secondary degree this implies an additional 12% of their time endowment on post-formal education activities.

²¹Effective tax rates are constructed following the methodology in Mendoza et al. (1994) more details on the methodology and the associated database are provided in Kostarakos and Varthalitis (2019).

85.3% which is consistent with EA-19 data over the period of fiscal stress between 2008-2014.²² Regarding public spending instruments we set transfers as a share of GDP equal to 25.4%.²³ Recall that as pointed out above we assume that transfers are unevenly distributed between the two household types favouring *Poor* households. Given the difficulty to retrieve the exact share allocated to each household type we assume that *Poor* households receive double the amount of transfers in nominal terms relative to *Rich* households. The rest of the public expenditure sub-components have been retrieved from Eurostat COFOG database which breaks down public spending per functional use. This provides an elegant disentanglement of total public spending into the main spending components of our model. For instance, public spending on education, s^{g^e} , is set at 1%, which is close to the post-formal schooling public spending on education.²⁴ For simplicity we assume that this share is equally allocated between the two household types, i.e. we set $\omega = 0.5$. Spending on public investment as a share of GDP, s^{g^i} , is set at 3.1%, based on data reported in the Economic Affairs function of the COFOG database. Finally, we allow the output share of government consumption, s^{g^e} , to be residually determined.²⁵

 $^{^{22}}$ Particularly we use the time series "General government consolidated gross debt - Excessive deficit procedure based on ESA 2010".

²³This includes the items of social benefits and other current transfers as those are defined in Eurostat National Accounts Indicators (ESA 2010).

²⁴This includes mainly government expenditure on tertiary education, post-secondary education other than tertiary and upper secondary education level.

²⁵In our model the definition of government consumption follows the COFOG database definition, and thus, includes a broad range of government functions such as general public services, public order and defence, recreation and culture, environmental protection, household and community amenities, health.

Parameter	Definition	Value			
	Households				
$0\leq\beta\leq 1$	time discount factor	0.978			
$0<\xi<1$	public consumption weight in composite consumption	0.100			
$\mu_1 > 0$	preference weight in the utility	0.400			
$\mu_2 > 0$	preference weight in the utility	0.600			
$0 \le \delta^k \le 1$	depreciation rate of physical capital	0.060			
$0 \le \delta^h \le 1$	depreciation rate of human capital	0.060			
$0 \leq \delta^g \leq 1$	depreciation rate of public capital	0.060			
$0 < n_R < 1$	population share of <i>Rich</i>	0.700			
$0 < n_P < 1$	population share of <i>Poor</i>	0.300			
	Production				
$0 < \alpha < 1$	share of composite input	0.980			
$\sigma < 1$	capital and skilled labor to unskilled labor substitution	0.323			
v < 1	capital to skilled labor substitution	-0.553			
0 < m < 1	labor share of <i>Rich</i>	0.200			
$0<\rho<1$	share of physical capital in the composite input	0.680			
$0 \leq \theta \leq 1$	elasticity of education time	0.800			
$0 \leq 1-\theta \leq 1$	elasticity of public education spending	0.200			
A > 0	scale parameter	1.000			
$B_R > 0$	human capital technology parameter of $Rich$	1.000			
$B_P > 0$	human capital technology parameter of <i>Poor</i>	1.000			
$0 < x_R < 1$	returns to scale for new human capital of $Rich$	0.400			
$0 < x_P < 1$	returns to scale for new human capital of <i>Poor</i>	0.350			

Table 1a: Parameter values

Instrument	Definition	Value			
	Tax rates				
τ^k	capital tax rate	0.368			
$ au^l$	labor tax rate	0.467			
$ au^c$	consumption tax rate	0.198			
Public spending					
s^{g^e}	GDP share of public education spending	0.010			
s^{g^c}	GDP share of government consumption	0.223			
s^{g^i}	GDP share of public investment	0.030			
s^{tr}	GDP share of government transfers	0.254			

Table 1b: Fiscal policy instruments

3.2 Status quo steady state solution

The steady state solution of the model, when we use the parameter values and the policy instruments of Tables 1a-1b, is reported in Table 2. In what follows, we refer to this steady state solution as the status quo economy and will serve as the point of departure for the fiscal policy experiments studied in the next sections. The implied numerical solution replicates some key macroeconomic ratios observed in the EA-19 like consumption as a share of output, physical capital as a share of output, debt as a share of output and skill premium.²⁶

²⁶Note that the values in Table 2 are defined as follows: $\bar{r} \equiv (1 - \tau^k)(r - \delta^k) \equiv r^b, \frac{c}{y} \equiv \frac{n_R c_R + n_R c_R}{n^f y}, \frac{k_R}{y} \equiv \frac{n_R k_R}{n^f y}, q \equiv \frac{n_R b_R}{n^f y}$.

Main variables		Model	
c_R	consumption of <i>Rich</i>	1.2494	
c_P	consumption of <i>Poor</i>	0.9327	
l_R	skilled labor	0.2444	
l_P	unskilled labor	0.2020	
e_R	time in education of <i>Rich</i>	0.1419	
e_P	time in education of <i>Poor</i>	0.1026	
\overline{r}	real return to physical capital	0.0224	
y	output	2.9556	
Key ratios		Model	Data
$\frac{c}{y}$	consumption as share of GDP	0.5580	0.5580
$rac{ec{k}_R}{y}$	physical capital as share of GDP	2.9600	2.9600
q	debt as a share of GDP	0.8530	0.8530
$\frac{w_R}{w_P}$	skill premium	1.5500	1.5500

Table 2: Status quo solution

4 Fiscal policy experiment

In this section, we define the fiscal policy experiments. The economy starts from its status quo steady state, as computed in Table 2, and travels towards a new reformed steady state with lower public debt-to-GDP ratio. Since public debt-to-GDP ratio is lower in the new reformed economy, the government can exploit the resulting fiscal space to increase public spending and/or reduce taxes. On the other hand, the opposite should happen in the transition to the new reformed steady state since such a debt reduction unavoidably requires cuts in spending and/or rises in taxes in the short and medium term.

Specifically, in the new reformed steady state, the debt-to-output ratio is reduced from 85.3%, which is the EA-19 data average over the fiscal stress period 2008-2014, to 60%; this exogenous reduction allows one spending (tax) instrument to increase (decrease) residually. Note that the 60% policy target is chosen simply to reflect the criteria set by the Maastrict Treaty. The government can achieve the transition from the status quo to the new reformed steady state by implementing different fiscal policy mixes following the fiscal feedback rules given by equations (60)-(67) in the Appendix C. We will compute constrained optimized fiscal policy rules so as our results will not depend on ad hoc differences in feedback policy coefficients. To do this, we follow Schmitt-Grohé and Uribe (2004, 2007) so that we limit our attention to fiscal rules that satisfy the following conditions. First, they are constrained, i.e. the fiscal instruments can react to a small number of easily observable macroeconomic indicators. Second, this reaction guarantees

local determinacy of the rational expectations equilibrium.²⁷ Third, they are optimized, i.e. policymakers choose the feedback policy coefficients to maximize a welfare criterion. The welfare criterion is the weighted conditional welfare of the *Rich* and *Poor* households as defined in (1) and (8) respectively, i.e. $W_0 = \tilde{n}V_{R,0} + (1 - \tilde{n})V_{P,0}$, where \tilde{n} denotes the weight assigned to *Rich* households' lifetime welfare. In our main experiments, we assume a Benthamite government meaning that we set the weights in the government's objective function equal to the associated population shares, i.e. $\tilde{n} = n_R$. Notice that welfare is computed conditional on the initial conditions which are given by the status quo solution as computed in Table 2.²⁸

5 Main Results

In this section, we present the aggregate and distributional implications of fiscal consolidation as defined above. Subsection 5.1 focuses on the reformed steady state, while, transition results are in subsection 5.2. To quantify aggregate effects, we will employ variables such as output, y, and social welfare, W, while, to measure distributional effects, we find it natural to use variables like the net income²⁹ of *Rich* and *Poor* households, y_R^{net} and y_P^{net} respectively, the relative net income (defined as the ratio of the *Rich* household's net income to the net income of the *Poor* household, $\frac{y_R^{net}}{y_P^{net}}$), the skill premium (which is typically defined as $\frac{w_R}{w_P}$), household specific welfares, V_R and V_P , and the relative welfare (defined as the ratio of the *Rich* household's discounted lifetime utility to the discounted lifetime utility of the *Poor* household, $\frac{V_R}{V_P}$).

5.1 Aggregate and distributional implications of debt consolidation in the reformed steady state

In this subsection, we present the new reformed steady state. As in e.g. Coenen et al. (2008) and Philippopoulos et al. (2015, 2017a, 2017b), once the debt-to-output ratio has been reduced, the government can increase spending or/and decrease taxes by taking advantage of the fiscal space created; this is the so called long run fiscal gain of debt consolidation.

Table 3 presents the key endogenous variables of the model capturing aggregate and distributional effects. In this table, we vary the residual fiscal instrument that adjusts in the long run and present the associated values of output, y, household specific net incomes, y_R^{net} and y_P^{net} , the relative **net** income, $\frac{y_R^{net}}{y_P^{net}}$, and skill premium, $\frac{w_R}{w_P}$. All values in Table 3 are reported

²⁷In addition, when necessary, we further restrict the vector of policy coefficients to lie within a space that delivers plausible values for fiscal instruments, i.e. $s^{g^c}, s^{g^i}, s^{g^e}, s^{tr_P}, s^{tr_P} > 0$ and $0 < \tau^c, \tau^k, \tau^n < 1$.

 $^{^{28}}$ In particular, we take a second order approximation to both the equilibrium conditions and the welfare criterion. First, we compute a second order approximation of both conditional welfare and the decentralized equilibrium around the reformed steady state as functions of the vector of feedback policy coefficients. Then, we use an optimization routine like *fminsearch* to compute the values of the feedback policy coefficients that maximize the conditional welfare criterion. For more details see Philippopoulos et al. (2017a) and (2017b). Dynare and Matlab routines are available upon request.

²⁹Net income is defined as gross income minus all types of taxes, i.e. net income of *Rich* households is $y_{R,t}^{net} \equiv (1 - \tau_t^k) (r_t k_{R,t} + \pi_t) + (1 - \tau_t^l) w_{R,t} l_{R,t} h_{R,t} + r_t^b b_{R,t} + \overline{tr}_{R,t} - \tau_t^c c_{R,t}$, and of *Poor* households is $y_{P,t}^{net} \equiv (1 - \tau_t^l) w_{P,t} l_{P,t} h_{P,t} + \overline{tr}_{P,t} - \tau_t^c c_{P,t}$.

as percentage deviations from their status quo values. A positive (negative) value implies an increase (decrease) vis-à-vis its status quo value. The first column of Table 3 reports which fiscal instrument adjusts in the new steady state taking advantage of the post-consolidation fiscal space, while, in the last column we compute the magnitude of the associated fiscal adjustment. The fiscal instruments which adjust in the new reformed economies are the output share of government consumption, s^{g^c} , the output share of government investment, s^{g^i} , the output share of public transfers to Poor households³⁰, s^{tr_P} , the output share of public spending on education, s^{g^e} , the tax rates on consumption, τ^c , labor income, τ^l , and capital income, τ^k . To understand the mechanisms of each reform, we experiment with one spending/tax policy instrument at a time keeping the others constant at their status quo values.

Fiscal Instr.	y	y_R^{net}	y_P^{net}	$\frac{y_R^{net}}{y_P^{net}}$	$rac{w_R}{w_P}$	$\Delta Inst$
s^{g^c}	0.0069	-0.0042	0.0058	-0.0100	-0.0077	0.0047
s^{g^i}	0.0124	0.0016	0.0111	-0.0095	-0.0069	0.0047
s^{tr_P}	0.0009	-0.0034	0.0128	-0.0162	-0.0129	0.0056
s^{g^e}	0.0279	0.0168	0.0261	-0.0093	-0.0120	0.0047
τ^{c}	0.0079	0.0052	0.0153	-0.0101	-0.0074	-0.0101
$ au^l$	0.0158	0.0137	0.0217	-0.0080	-0.0051	-0.0095
$ au^k$	0.0175	0.0119	0.0150	-0.0031	-0.0005	-0.0133

Table 3: Steady state output and distributional effects in the various reformed economies (as % deviations from status quo)

A key message from Table 3 is that debt consolidation is always output enhancing and, perhaps more interestingly, progressive in the long run. That is, output, y, increases (see column 2 in Table 3), while relative net income (income inequality), $\frac{y_R^{net}}{y_P^{net}}$, and skill premium (wage inequality), $\frac{w_R}{w_P}$, decrease (see column 4 and 5 in Table 3 respectively) vis-a-vis their status quo value in all reformed economies. Thus, fiscal consolidation can improve both equity and efficiency in the long run.

Regarding output, debt consolidation is more productive in the long run when the government increases its productive spending (like public spending on education or investment) or reduces distortionary taxation. Actually, the best instrument to use in terms of output, as well as in terms of net income of all individuals, is public spending on education.

The relative income, $\frac{y_R^{net}}{y_P^{net}}$, decreases in all reformed economies which means that debt consolidation benefits more the *Poor* households. This is mostly driven by the sharp decrease in the income coming from government bonds earned by *Rich* households which gets lower under debt consolidation.³¹ The reduction in the net income of *Rich* households is less striking when the government reduces income taxes (see the last two rows in Table 3) for two reasons. First,

³⁰ To save space, we do not report results for public transfers to *Rich* household, s^{tr_R} . Note that for *Rich* households the Ricardian equivalence holds, thus, the long run effects are trivial.

³¹These findings are in line with Schwarzmüller and Wolters (2015). However, they focus on consumption inequality rather than net income and wages.

lower income taxes imply higher wealth; recall that *Rich* households earn capital income while they receive higher wages than *Poor* households.³² Second, the resulting increase in output requires additional physical capital and, due to the complementarity between physical capital and skilled labor, more skilled than unskilled labor, which again favors the *Rich* households. These effects moderate the adverse effect on the net income of *Rich* households from the decrease in income coming from bond holdings. In terms of wage inequality (see column 4 in Table 3), debt consolidation always reduces the skill premium in the long run.³³

Similar studies have examined the effects of fiscal consolidation on aggregate welfare (see e.g. Bi and Kumhof (2011) and Philippopoulos et al. (2017a) and (2017b)). To this end, in Table 4, we compute the percentage deviations of social welfare, W, Rich and Poor households' discounted lifetime steady state welfare, V_R and V_P respectively, as well as the relative welfare, $\frac{V_R}{V_P}$.

Fiscal	W	V	U.	V _R
Instr.	VV	VR	VP	$\overline{V_P}$
s^{g^c}	-0.0077	-0.0225	0.0174	-0.0407
s^{g^i}	0.0042	-0.0089	0.0264	-0.0363
s^{tr_P}	0.0047	-0.0158	0.0394	-0.0574
s^{g^e}	0.0781	0.0769	0.0801	-0.0035
$ au^c$	0.0131	0.0009	0.0337	-0.0339
$ au^l$	0.0235	0.0145	0.0387	-0.0252
$ au^k$	0.0271	0.0214	0.0368	-0.0161

Table 4: Steady state social and individual life-time welfare (as % deviations from status quo³⁴)

Welfare results in Table 4 are consistent with the findings in Table 3. In particular, fiscal consolidation enhances social welfare and also reduces the welfare gap between *Rich* and *Poor* households in all reformed economies. Similarly to the results in Table 3, the increase in government spending on education yields the highest welfare gains both for the society and for each income class of households. Also, using the fiscal space created by debt consolidation to finance an increase in spending on public education and/or to finance a cut in distortionary taxation

 $^{^{32}}$ Table E.1 in Appendix E presents the various sources of income by household type as percentage deviations from their status quo value.

³³The net effect on skill premium depends on which of the capital-skill complementarity or the relative skill supply effect dominates. On the one hand, the increase in output requires more physical capital pushing skill premium upwards. On the other hand, relative skill supply increases pushing skill premium in the opposite direction. In our experiments the latter effect is stronger.

³⁴In Table 4 a positive (negative) value denotes a welfare improvement (deterioration) vis-à-vis its status quo value. Since we assume logarithmic utility function, welfare takes negative values, i.e. $W, V_R, V_P < 0$. For illustrative purposes, we define the percentage deviations of, say social welfare measure, from its status quo value as $\frac{(W^{\phi}) - (W^{SQ})}{|W^{SQ}|}$, where W^{ϕ} and W^{SQ} denote social steady state welfare in any reformed economy and status quo economy respectively. In contrast the relative welfare is a positive number, $\frac{V_R}{V_P} > 0$. In Table 4, we report $\frac{(\frac{V_R}{V_P})^{SQ} - (\frac{V_R}{V_P})^{\phi}}{(\frac{V_R}{V_P})^{SQ}}$. As before, a positive (negative) value means that the welfare gap increases (decreases).

is Pareto efficient, i.e., both *Rich* and *Poor* households get better off in the reformed economy. On the other hand, fiscal consolidation is not always Pareto efficient. For example, although increases in government consumption, public transfers targeted to *Poor* households and public investment can also enhance the aggregate output, while, the last two also improve the social welfare, they are not Pareto efficient. Any small social welfare gains arise solely because of the increase in *Poor* households' welfare; the results in Table 4 show that the welfare of the *Rich* is reduced.

5.2Aggregate and distributional implications of debt consolidation along the transition

In this section, we focus our analysis on the transition implications of public debt consolidation. The economy departs from its status quo steady state and moves towards a new reformed economy with lower debt to GDP ratio equal to 60%.³⁵ This requires fiscal policy to use one (or more) fiscal instruments to react to debt deviations from its new target. We focus our analysis on constrained optimized fiscal policy rules as analyzed in section 4.

The optimized policy mix suggest that fiscal policy should cut government consumption sharply to consolidate its debt while at the same time keep constant distortionary taxation and productive public spending. The resulting optimized values of feedback policy coefficients reported in the note of Table 5 suggest that increasing distortionary income taxation and/or decreasing productive spending to reduce public debt is not recommended. These results are consistent with findings in similar studies see in Philippopoulos et al. (2015) and (2017b).

Table 5 computes the average percentage deviations from their status quo value of output, y_{i} relative net income, $\frac{y_R^{net}}{y_R^{net}}$, and skill premium, $\frac{w_R}{w_P}$, over various time horizons under the constrained optimized fiscal policy.³⁶

Variable	2 years	4 years	6 years	10 years
y	0.002	0.013	0.0155	0.016
$rac{y_R^{net}}{y_P^{net}}$	-0.037	-0.033	-0.031	-0.027
$rac{w_R}{w_P}$	0.0443	0.0277	0.0208	0.0138
Noto: Th	o constrain	d optimized	policy cooffic	ionts aro

Table 5: Output, net income ratio and skill premium over various time horizons with optimized policy

The constrained optimized policy coefficients are

 $\gamma_q^{g^c} = 0.7, \ \gamma_q^c = \gamma_q^l = \gamma_q^k = \gamma_q^{g^e} = \gamma_q^{g^i} = \gamma_q^{tr_R} = \gamma_q^{tr_P} = 0$

³⁶Table 5 presents results from model based simulations generated by the first order accurate approximation of the equilibrium system.

³⁵To save space, we present results for the transition to a reformed economy in which government consumption is the fiscal instrument that reaps the benefit of debt reduction. In section 7.3 we present the associated transitional dynamics when the economy travels towards the reformed economies where public spending on education or capital tax are the residual fiscal instruments. Results for the rest of the reformed economies reported in Table 3 are available upon request.

Results in Table 5 imply that fiscal consolidation is efficient in the medium and long run meaning that aggregate output, y, increases above its pre-reform value after the second year. Also, the relative net income, $\frac{y_R^{net}}{y_P^{net}}$, is reduced over all time horizons. This is attributed to the sharp reduction of the *Rich* households' income coming from bonds over all time horizons and to the temporary reduction in their capital income. However, wage inequality, $\frac{w_R}{w_P}$, widens for a prolonged period. As we also discuss below, debt consolidation results in an increase in physical capital stock which due to the capital-skill complementarity channel in our model increases skill premium.

Similarly, Table 6 presents social welfare and household specific welfare as percentage deviations from a reference regime³⁷ under the constrained optimized fiscal consolidation policy over various time horizons.³⁸ A positive value implies that the associated welfare criterion increases vis-à-vis the reference regime.

Table 0.	wenare over	various time	TIONZONS WIT	ii optimized policy
Welfare	2 years	4 years	6 years	10 years
Social	0.018	0.0383	0.0538	0.0736
Rich	0.0354	0.0558	0.0704	0.087
Poor	-0.0236	-0.0027	0.0150	0.0421
	Not	te: Feedbacks	as in Table 5.	

Table 6: Welfare over various time horizons with optimized policy

Welfare results in Table 6 suggest that fiscal consolidation is social welfare enhancing over all time horizons (see second row), however, it does not produce a Pareto efficient outcome in the short run. In particular, in the first four years this reform hurts *Poor* households and benefits *Rich* households. As can be seen in Table 6, *Poor* households' welfare is reduced over the first four years (see fourth row), while, *Rich* households welfare increases in short/medium run (see third row). That is debt consolidation is Pareto efficient only in the medium and long run. To understand the logic of the results in Tables 5 and 6 we discuss the underlying transmission mechanism in the next section.

6 The underlying transmission mechanism

In this section we present the dynamic paths of the key endogenous variables of the model during the transition phase of fiscal consolidation. This provides insight into the transmission channel of fiscal consolidation policy in our economy and allows us to understand the logic of our results in Section 5. As before, the economy departs from its status quo steady state and travels to one of the reformed economies reported in Section 5.1. Figure 1 depicts the dynamic paths of the

 $^{^{37}}$ We can use multiple reference regimes, for simplicity and comparability with results in Table 5, we compute percentage deviations from a reference regime in which the economy stays at its status quo steady state forever.

³⁸Table 6 presents results from model-based simulations generated by the second order accurate approximation of the equilibrium system and the welfare criteria.

main endogenous variables under the constrained optimized fiscal policy, while, in the long run, the government uses the resulting fiscal space to increase government consumption.



Figure 1: Dynamic paths of the main endogenous variables with optimized policy

Notes: Endogenous variables are expressed as percentage deviations from their status quo value. Total consumption is defined as $c \equiv n_R c_R + n_P c_P$.

We start with aggregate effects. Fiscal consolidation implemented via the constrained optimized fiscal policy (i.e. via government consumption cuts³⁹) decreases aggregate demand which induces a temporary output contraction (see the dynamic path of y). Subsequently, the debt adjustment and the sharp reduction in government consumption cause a reduction in real interest rates, r^b and r. This crowds in total private consumption, c, and investment, i_R , whereas the latter results in the rise of the physical capital stock; see the dynamic path of k_R . Thus, output recovers above its pre consolidation value and converges to its new higher value in the reformed steady state. The increase in the capital stock is significant and prolonged as the economy converges to a reformed steady state with higher capital stock.

We now turn to the distributional implications of fiscal consolidation. First, due to the capital-skill complementarity channel, the rise in physical capital induce an increase in the wages of *Rich* (skilled) households,⁴⁰ see the dynamic path of w_R . Thus, the skill premium (wage

 $^{^{39}}$ To save space we present the dynamic paths of fiscal policy instruments and other endogenous variables in Figure E.1 in the Appendix E.

 $^{^{40}}$ At the same time, *Rich* households substitute labor with leisure in the very short run (see Figure E.1 in the Appendix E) and thus relative skill supply, $\frac{l_{R,t}h_{R,t}}{l_{P,t}h_{P,t}}$ decreases in equilibrium. Both relative skill supply and capital skill complementarity effects push upward skill premium.

inequality), $\frac{w_R}{w_P}$, increases in the transition phase of fiscal consolidation. On the other hand, relative net income, $\frac{y_R^{net}}{y_P^{net}}$, falls over the transition to the reformed steady state and converges to its new lower value.⁴¹The relative net income is a function of various endogenous variables of the model, thus, to understand its dynamic path, we decompose it into the various sources of income of *Rich* and *Poor* households. Results in Figure 1 indicate that the decrease in the relative income can be mostly attributed to the sharp decrease in the income coming from government bonds (see the dynamic paths of $r^b b_R$), the temporary reduction in the capital income due to the fall in the return on capital (see the dynamic path of r and rk_R respectively) and the higher income taxes paid by *Rich* households.⁴² On the other hand, the labor income of *Rich* households, $w_R l_R h_R$, increases relatively more than the labor income of *Poor* households, $w_P l_P h_P$, due to the capital-skill complementarity effect.

Finally, to understand the welfare results in Table 6, we focus on the variables that enter the utility functions of *Rich* and *Poor* households. Consumption of *Rich* households, c_R , increases due to the crowding in effects induced by debt consolidation and the sharp fall in interests rates. Also, since the labor income of *Rich* households increases (mostly due to the increase in their wages), these households substitute hours worked with leisure. Both effects enhance the welfare of *Rich* households in the short/medium run.

On the other hand, debt consolidation puts downward pressure to the wages of *Poor* (unskilled) households, w_P , and thus they reduce consumption, c_P , on impact; recall that they cannot smooth their consumption over time. In addition, in the short run, due to lower wages, they need to increase their labor supply to boost their labor income. Both consumption and labor dynamics contribute to the welfare losses experienced by *Poor* households in the short run.

7 Sensitivity analysis

In this section we conduct a sensitivity analysis. In Sections 7.1 and 7.2, we focus on the key sources of inequality in our model to shed light on the different channels through which they affect our main results. In section 7.3, we discuss transitional dynamics towards alternative reformed economies. In Section 7.4, we discuss similarities of constrained optimized fiscal policy with a Ramsey-type policy in our model. Finally, Section 7.5 reports sensitivity tests with respect to other key structural parameters of the model. In all the exercises conducted in this section, we compute the constrained optimized fiscal policy rules as stated in Section 4 under the new calibration and/or modelling changes. Here, we report that the optimized policy mix remains the same.

⁴¹In an earlier version of this paper we show that income inequality can increase in the short run under an ad hoc (non optimized) spending based fiscal consolidation policy see in Sakkas and Varthalitis (2018).

⁴²Although government imposes identical tax rates to both households, distortionary taxes are analogous to income. This implies that a relatively higher income increase is accompanied with higher taxes.

7.1 Capital-skill complementarity

In this section we focus on the structural parameter that governs the capital-skill complementarity (CSC), i.e. v < 1, in equation (13) in section 2.5. This parameter governs the elasticity of substitution between physical capital and skilled labor, $\frac{1}{1-v}$. A smaller (larger) value of v(i.e. a smaller (larger) value of $\frac{1}{1-v}$) corresponds to a higher (lower) degree of complementarity between skilled labor and physical capital.

Figure 2 presents the dynamic paths of the key endogenous variables of the model in three economies when we vary the degree of capital-skill complementarity. First, the blue lines present the dynamic responses from Section 6 (we include these for comparison); recall that the value of v in the benchmark calibration is equal to -0.553 (which means $\frac{1}{1-v} = 0.6439$). Second, the yellow lines present the dynamic responses in an economy which exhibits lower capital-skill complementarity (labeled as low-CSC), for example we set v = 0.1 > -0.55. Notice that this value implies a higher elasticity of substitution between capital and skilled labor with respect to the benchmark economy, i.e. $\frac{1}{1-v} = 1.11 > 0.6439$. Third, the red lines simulate an economy which exhibits higher capital-skill complementarity (labeled as high-CSC), for example we set v = -1 < -0.55. Similarly, this value implies a lower elasticity of substitution between capital and skilled labor with respect to the benchmark economy, i.e. $\frac{1}{1-v} = 1.11 > 0.6439$. Third, the red lines simulate an economy which exhibits higher capital-skill complementarity (labeled as high-CSC), for example we set v = -1 < -0.55. Similarly, this value implies a lower elasticity of substitution between capital and skilled labor with respect to the benchmark economy, i.e. $\frac{1}{1-v} = 0.5 < 0.6439$.

Figure 2: Dynamic paths of the main endogenous variables under various degrees of capital-skill complementarity



Notes: See notes in Figure 1.

Comparing the dynamic paths of aggregate output, y, relative income, $\frac{y_R^{net}}{y_P^{net}}$, and skill premium, $\frac{w_R}{w_P}$, we infer that debt consolidation's aggregate and distributional effects remain qualitatively similar across the three economies (compare benchmark, blue lines, with high-CSC, red lines, and low-CSC, yellow lines).

However, as expected, the capital-skill complementarity channel plays a key role in the dynamics of wage inequality (see the dynamic path of $\frac{w_R}{w_P}$ across the three economies). A larger degree of capital-skill complementarity (i.e. lower elasticity of substitution between capital and skilled labor, $\frac{1}{1-v}$) favors *Rich* households in the early phase of fiscal consolidation in terms of wage and labor income. Particularly, as the degree of capital-skill complementarity increases, the wage of *Rich* households, w_R , increases more, while the wage of *Poor* households, w_P , decreases relatively more and thus wage inequality, $\frac{w_R}{w_P}$, widens. Similarly, the increase in the labor income of *Rich* households, $w_R l_R h_R$, is larger, while, the increase in the labor income of *Poor* household, $w_P l_P h_P$, is smaller as we increase the degree of capital skill complementarity.

The logic of these results is the following. A larger degree of capital-skill complementarity (i.e. a lower value of v) in the production function results in, *ceteris paribus*, a larger reduction in the marginal product of capital, see the dynamic path of r. This subsequently leads to a larger rise in the physical capital stock in the short run, k_R , and thus, due to the higher capital-skill complementarity, a higher rise in the wage of *Rich* (skilled) households, w_R .

On the other hand, the short run reduction in the relative income, $\frac{y_R^{net}}{y_P^{net}}$, is relatively bigger (smaller) on impact in the high (low)-CSC economy, while, in the medium run, the dynamic paths are almost identical with the one of the benchmark economy. This mostly happens due to the larger (smaller) reduction in real rates, see the dynamic paths of r and r^b , which imply a larger (smaller) decrease in capital income, rk_R , and income coming from government bonds, $r^b b_R$, for *Rich* households.

7.2 Access to capital/financial markets

In this section we relax the assumption that *Poor* households have no access to financial/capital markets. In particular, we allow both *Rich* and *Poor* households to accumulate physical capital and save in the form of government bonds subject to transactions costs as in e.g. Angelopoulos et al. (2014). The role of transactions costs is twofold, first to guarantee a well defined steady state solution and ensure stationarity in the transition, second to allow us to maintain heterogeneity in asset holdings between *Rich* and *Poor* households. To do this, we assume that *Poor* households face consistently higher transactions costs than *Rich* households which implies that *Rich* households have consistently higher wealth than *Poor* households. To save space we present modelling details in Appendix D. For illustrative purposes, here we report that the associated parameters which measure the size of these transactions costs for capital and government bonds are denoted as χ_R , ϕ_R and χ_P , ϕ_P for *Rich* and *Poor* households respectively.

To understand the role of asset heterogeneity in our main results, in this section, we present results from three economies varying the degree of financial constraints faced by *Poor* households. As we increase the magnitude of the associated transactions costs for *Poor* households, asset heterogeneity increases since the *Poor* reduce their asset holdings and rely more on labor income.

In Figure 3, first, we simulate an economy in which we allow *Poor* households to accumulate physical capital and save in the form of government bonds. As said, to maintain asset heterogeneity they are subject to relatively higher transaction costs than *Rich* households, for example we set $\chi_R = \phi_R = 0.001 < \chi_P = \phi_P = 0.01$. Second, we simulate an economy in which we further increase asset heterogeneity between the *Rich* and the *Poor*. That is *Poor* households can still accumulate physical capital and save in the form of government bonds but now are subject to much higher transaction costs than *Rich* households, for example we set $\chi_R = \phi_R = 0.001 < \chi_P = \phi_P = 0.1$. The latter economy is closer to the economy presented in Section 6, where *Poor* households live hand to mouth. Finally, for comparison, we also include results from Section 6.

Figure 3 presents the dynamic responses of the key endogenous variables in the three economies under the constrained optimized fiscal rules.

Figure 3: Dynamic paths of the main endogenous variables under various degrees of asset heterogeneity



Notes: See notes in Figure 1. Total physical capital and private investment are $k \equiv n_R k_R + n_P k_P$ and $i \equiv n_R i_R + n_P i_P$ respectively. We set $\chi_R = \phi_R = 0.001$ in the economies represented with red and yellow lines.

Figure 3 shows that the degree of asset heterogeneity does not alter significantly the aggregate effects of fiscal consolidation, however, as expected, it affects its distributional implications.

Perhaps, a rather novel finding is that the degree of asset heterogeneity affects significantly wage inequality $\frac{w_R}{w_P}$. That is the higher the transactions costs faced by *Poor* households, i.e. the larger the inequality in asset distribution, the larger the increase in wage inequality in the short run phase of fiscal consolidation. This is apparent when we compare the impact response of $\frac{w_R}{w_P}$ in the benchmark economy, where, *Poor* households have zero access in financial markets (blue lines) with the economies where we allow them partial access to financial markets subject to high/low transactions costs (see red/yellow lines respectively).

In the latter economies, the negative effect of debt consolidation on *Poor* households is less severe since they do not solely rely on labor income. As a result they do not need to increase labor supply and substitute leisure with labor to the same extent as in the benchmark economy. For that reason, the equilibrium wage of *Poor* households, w_P , increases in the short run which leads to a smaller increase in wage inequality vis-à-vis the benchmark economy.

Finally, as expected, the relative net income, $\frac{y_R^{net}}{y_P^{net}}$, reduces less as we decrease asset heterogeneity. Now, since *Poor* households earn income from government bonds and capital, their income decreases analogously to the income of *Rich* households coming from the same sources (due to reasons discussed in Section 6) causing a smaller reduction in the relative income (income inequality).

7.3 Transition towards alternative reformed economies

In section 5.1, we showed that the long run aggregate and distributional effects of debt consolidation depend on the fiscal instrument which adjusts in the new reformed steady state to exploit the post consolidation fiscal space. So far in our analysis of the transmission mechanism, we present results from the case in which fiscal policy adjusts government consumption in the new reformed steady state. In Figure 4, we present the dynamic paths of key endogenous variables of the model when the economy departs from the status quo steady state, as above, and converges to two of the alternative reformed economies discussed in Section 5.1. That is, fiscal policy implements the constrained optimized fiscal policy rules computed as in Section 4, while, the economy converges to two alternative reformed steady states, namely, where the government increases public spending on education, s^{g^e} , or decreases capital tax, τ^k . For notational convenience, in this section we label each of these policy experiments using the fiscal instrument utilized in the new reformed steady state.

As above, for comparison, the blue lines present the dynamic responses of Section 6 (labeled as s^{g^c}). In addition, the red lines present the dynamic responses for the case in which the government exploits the fiscal space to increase public spending on education in the new reformed steady state (labelled as s^{g^e}). Finally, yellow lines present the dynamic responses for the case in which the government cuts capital taxes in the new reformed steady state (labelled as τ^k). We choose these reformed economies simply because they deliver the best steady state welfare and output outcomes (see Tables 3 and 4 in Section 5.1).

Figure 4: Dynamic paths of the main endogenous variables when the economy converges to alternative reformed economies



Notes: See notes in Figure 1.

We report that the constrained optimized fiscal policy does not depend on the reformed steady state at which the economy travels to. Also, Figure 4 indicates that our key results analyzed in Section 6 remain the same. However, some new results are worth mentioning. First, the larger long run aggregate benefit of fiscal consolidation, arising from a more efficient allocation of the post consolidation fiscal space, can be brought forward in the short/medium run. This is reflected in the dynamic paths of endogenous variables like output, aggregate consumption and physical capital. Regarding output, y, the recession on impact is smaller, while, the rebound in the medium run is larger when the fiscal **space** is used more efficiently. In addition, the increase in aggregate consumption, c, and physical capital, k_R , are larger and more persistent when the economy travels to the reformed economies studied in this section (compare the blue lines, s^{g^c} , with the red, s^{g^e} , and yellow lines, τ^k).

This effect works through the expectation channel, that is, households and firms anticipate the positive effects from the increase in productive spending on education or the reduction in capital taxes and adjust their behaviour in the short run. However, investment in physical and human capital take time to materialize, and thus the positive effects are more evident in the medium/long run.

Short run dynamic responses in Figure 4 are also consistent with findings in Section 5.1. That is, as highlighted in Section 5.1, using the fiscal space created by debt consolidation to finance an increase in spending on education or to finance a cut in distortionary capital tax yields relatively better outcomes than the benchmark economy for both *Rich* and *Poor* households in

the short and medium run as well as in the long run. For example, *Rich* households experience a larger increase in their capital income in the medium run (see rk_R) and labor income in the short/medium run (see $w_R l_R h_R$) and, at the same time, *Poor* households also experience a larger increase in their labor income (see $w_P l_P h_P$).

A clear cut message is that using the fiscal space to finance policies that induce long run positive supply side effects benefits both households.

7.4 Discussion of the Ramsey approach

In this paper, in accordance with most of the related literature on debt consolidation (see the Introductory section), we have focused on constrained optimized state-contingent policy rules. That is, the values of policy instruments over time can deviate from their trend values by reacting optimally to a number of macroeconomic indicators, but these trend values are set as in the data averages rather than chosen optimally. Alternatively, one could compute Ramsey-type optimal policy, meaning completely optimal state-contingent policy rules. Our approach can by justified by the fact that institutional and political constraints do not allow for a fully optimal Ramsey-type policy, especially when one considers fiscal policy (see also e.g. Schmitt-Grohé and Uribe, 2005, Kirsanova et al., 2007, Philippopoulos et al., 2017b). Nevertheless, it is worth pointing out that the policy message coming from our constrained optimized rules is in accordance with the typical Ramsey recipe. Namely, under both approaches, the optimal design of fiscal policy takes the form of higher productivity-enhancing spending and/or lower capital taxes in the long run, while, in the short, any required increase in revenues should be achieved by changes in the least distorting fiscal instrument which, in our model (as is common in the debt consolidation literature), is cuts in non-productive public spending.

7.5 Other parameters

We conduct sensitivity analysis with respect to other structural parameters of the model, namely the degree of complementarity/substitutability between private and public consumption, $0 < \xi < 0.2$, human capital technology parameters, $0.9 < B_P \leq B_R < 1.1$, parameters that govern the returns to scale for *Rich* and *Poor* human capital, $0.25 < x_P \leq x_R < 0.5$, the weights in the government's objective function, i.e. $0.5 < \tilde{n} < 0.8$. We report that our main results do not change (the full set of results is available upon request from the authors).

8 Conclusions and possible extensions

In this paper, we assessed the aggregate and distributional implications of fiscal consolidation policy using a dynamic general equilibrium model with heterogeneous agents. Since the main results have been summarized in the Introduction, we close with some possible extensions.

A possible extension is to depart from the closed economy setup and study similar questions in an open economy model allowing for international mobility of capital and labor (i.e. migration). This could leave room for studying cross-border effects. Potential frictions in international capital and labor mobility could be an additional source of heterogeneity among different types of households. Thus, fiscal consolidation policy could affect aggregate and distributional outcomes through additional channels. We leave these extensions for future work.

References

- Almeida, V., G., Castro, R.M., Felix, and J.R. Maria. 2013. Fiscal consolidation in a Small Euro-Area Economy. *International Journal of Central Banking*, 9(4): 1-38.
- [2] Angelopoulos K., Benecchi A. and J. Malley. 2017. Can Subsidising Job-Related Training Reduce Inequality?. CESifo Working Paper Series No. 6605.
- [3] Angelopoulos, K. Jiang, W., and J. Malley. 2013. Tax reforms under market distortions in product and labour markets. *European Economic Review*, 61, issue C, pp.28-42.
- [4] Angelopoulos K., Malley J., and A., Philippopoulos. 2017. Human capital accumulation and transition to skilled employment. *Journal of Human Capital*, 11(1),pp.72-105.
- [5] Angelopoulos, K., Fernandez, B.X. and J.R. Malley. 2014. The Distributional Consequences of Tax Reforms Under Capital-Skill Complementarity. *Economica*, 81(324), 747-767.
- [6] Ardagna, S. 2007. Fiscal policy in unionized labor markets. Journal of Economic Dynamics and Control, 31(5), 1498-1534.
- Benigno, P. 2009. Price stability with imperfect financial integration. Journal of Money Credit and Banking. 41, 121-49
- [8] Bilbiie, F.O., Monacelli, T., and R., Perotti. 2013. Public debt and redistribution with borrowing constraints. *The Economic Journal*, 123(566), F64-F98.
- [9] Bi, H., and M. Kumhof. 2011. Jointly optimal monetary and fiscal policy rules under liquidity constraints. *Journal of Macroeconomics*, 33, 373-389.
- [10] Bi, H., Leeper, E., and C. Leith. 2013. Uncertain Fiscal Consolidations. The Economic Journal 123 (566), F31-F63.
- [11] Blankenau, W.F. and N.B. Simpson. 2004. Public education expenditures and growth. Journal of Development economics, 73(2), pp.583-605.
- [12] Cantore, C., Levine, P., Melina G., and J., Pearlman. 2012. Optimal fiscal and Monetary Rules in Normal and Abnormal times. *School of Economics Discussion Papers 0513*, School of Economics, University of Surrey.
- [13] Cardani, R., Menna, L., and P. Tirelli. 2018. The optimal policy mix to achieve public debt consolidation. *Macroeconomic Dynamics*, 1-17.
- [14] Chari V. V., Kehoe P. J., and E. R. McGrattan, 2007. Business Cycle Accounting. *Econo-metrica*, Econometric Society, vol. 75(3), 781-836, May
- [15] Coenen, G., Straub, R., and M. Trabandt. 2013. Gauging the effects of fiscal stimulus packages in the euro area. *Journal of Economic Dynamics and Control*, 37(2), 367-386.

- [16] Coenen, G., Mohr, M., and Straub R. 2008. Fiscal consolidation in the euro area: long run benefits and steady state costs. *Economic Modelling*, 25, no. 5, 912-932.
- [17] Crivellaro, E. 2016. The College Wage Premium over Time: Trends in Europe in the Last 15 Years, *Research in Labor Economics, in: Inequality: Causes and Consequences*, volume 43, 287-328 Emerald Publishing Ltd.
- [18] Cogan, J., Taylor, J., Wieland, V., and M.Wolters. 2013. Fiscal consolidation Strategy. Journal of Economic Dynamics and Control, 37 (2), 404-21.
- [19] Daniel, B.C. and S. Gao. 2015. Implications of productive government spending for fiscal policy. *Journal of Economic Dynamics and Control*, 55, pp.148-175.
- [20] Dolado, J.J., Motyovszki, G., and E. Pappa. 2018. Monetary Policy and Inequality under Labor Market Frictions and Capital-Skill Complementarity, CEPR Discussion Papers 12734, C.E.P.R. Discussion Papers.
- [21] Economides, G., Papageorgiou D., and A. Philippopoulos. 2017. The Greek Great Depression: a general equilibrium study of its drivers. Working papers 234, Bank of Greece.
- [22] Economides, G., Philippopoulos, A., and V. Vassilatos. 2014.Public, or private, providers of public goods? A dynamic general equilibrium study, *European Journal of Political Econ*omy, Elsevier, vol. 36(C), 303-327.
- [23] Economides, G., Papageorgiou, D., Philippopoulos A., and V., Vassilatos. 2012. Smaller public sectors in the euro area: Aggregate and distributional implications. *CESifo Economic Studies*, 59(3), 536-558.
- [24] Erceg, C. and J. Lindé. 2013. Fiscal consolidation in a currency union: spending cuts vs. tax hikes. Journal of Economic Dynamics and Control, 37, 422-445.
- [25] European Central Bank. 2017. ECB Forum on Central Banking: Investment and growth in advanced economies, Conference proceedings, pp. 333-346.
- [26] European Commission. 2017. Public Finances in the EMU.
- [27] European Commission. 2014. Public Finances in the EMU.
- [28] Forni, L., Gerali, A., and M., Pisani. 2010. The macroeconomics of fiscal consolidation in euro area countries. *Journal of Economic Dynamics and Control*, 34, 1791-1812.
- [29] Forni, L., Monteforte L., and L. Sessa. 2009. The general equilibrium effects of fiscal policy: Estimates for the euro area. *Journal of Public Economics*, 93(3-4), 559-585.
- [30] Furceri, D., Jalles, J.T., and P. Loungani. 2015. Fiscal consolidation and inequality in advanced economies: How robust is the link. Inequality and the role of fiscal policy: trends and policy options, IMF, Fiscal Affairs Department, Washington, DC, USA.

- [31] Gali, J., Lopez Salido, J. D., and Valles, J. 2007. Understanding the effects of government spending on consumption. *Journal of the European Economic Association*, 5(1), 227-270.
- [32] Garcia-Milà, T., Marcet, A., and E. Ventura. 2010. Supply side interventions and redistribution. *The Economic Journal*, 120(543), 105-130.
- [33] Glomm, G. and B. Ravikumar. 1997. Productive government expenditures and long run growth. Journal of Economic Dynamics and Control, 21(1), 183-204.
- [34] Gomes, P. 2018. Heterogeneity and the public sector wage policy. International Economic Review, forthcoming
- [35] Guo, J.T. and K.J. Lansing. 1997. Tax structure and welfare in a model of optimal fiscal policy. *Economic Review-Federal Reserve Bank of Cleveland*, 33, 11-23.
- [36] He, H. and Z. Liu. 2008. Investment-specific technological change, skill accumulation, and wage inequality. *Review of Economic Dynamics*, 11(2), 314-334.
- [37] Heathcote J., Perri F., G. L. Violante. 2010. Unequal We Stand: An Empirical Analysis of Economic Inequality in the United States: 1967-2006. *Review of Economic Dynamics*, Elsevier for the Society for Economic Dynamics, vol. 13(1), pages 15-51, January.
- [38] Kostarakos I. and Varthalitis P. 2018. Effective Tax Rates in Eurozone: An updated database over 1995-2017. ESRI working papers (forthcoming).
- [39] Kirsanova, T., Satchi, M., Vines, D., and S. Wren-Lewis. 2007. Optimal Fiscal Policy Rules in a Monetary Union. Journal of Money, Credit and Banking, 39 (7): 1759–84
- [40] Krueger D., Perri F., Pistaferri L., and G. L. Violante. 2010. Cross Sectional Facts for Macroeconomists. *Review of Economic Dynamics*, Elsevier for the Society for Economic Dynamics, vol. 13(1), pages 1-14, January.
- [41] Krusell, P., Ohanian, L. E., Ríos-Rull, V., and G. L. Violante. 2000. Capital-skill complementarity and inequality: A macroeconomic analysis. *Econometrica*, 68(5), 1029-1053.
- [42] Leeper, E.M. 2010. Monetary science, fiscal alchemy (No. w16510). National Bureau of Economic Research.
- [43] Lindquist, M., J. 2004. Capital-Skill Complementarity and Inequality Over the Business Cycle. *Review of Economic Dynamics*, Elsevier for the Society for Economic Dynamics, vol. 7(3), 519-540, July
- [44] Lucas, R. Jr. 1988. On the mechanics of economic development. Journal of Monetary Economics, Elsevier, vol. 22(1), pages 3-42, July
- [45] Maliar L., Maliar., S., and I. Tsener. 2017. Capital-Skill Complementarity and Inequality: Twenty years after, *Mimeo*

- [46] Mendoza, E. G., Razin A., and L. L. Tesar. 1994. Effective tax rates in macroeconomics: Cross-country estimates of tax rates on factor incomes and consumption, *Journal of Mon*etary Economics, Elsevier, vol. 34(3), 297-323, December.
- [47] Michaud, A. and J. Rothert. 2018. Redistributive fiscal policies and business cycles in emerging economies. *Journal of International Economics*, 112, 123-133.
- [48] OECD. 2018. Education at a Glance 2018: OECD Indicators, OECD Publishing, Paris
- [49] Pappa, E., Sajedi, R., and E.Vella. 2015. Fiscal consolidation with tax evasion and corruption. Journal of International Economics, vol. 96(1), 56-75.
- [50] Philippopoulos A., Varthalitis, P., and V. Vassilatos. 2015. Optimal Fiscal and Monetary Policy Action in a Closed Economy. *Economic Modelling*, 48, 175-88.
- [51] Philippopoulos A., Varthalitis, P., and V. Vassilatos. 2017a. Fiscal consolidation and its cross-country effects. *Journal of Economic Dynamics and Control*, 83, 55-106.
- [52] Philippopoulos A., Varthalitis, P., and V. Vassilatos, 2017b. Fiscal Consolidation in an Open Economy with Sovereign Premia and without Monetary Policy Independence". *International Journal of Central Banking* 13, 259-306.
- [53] Roubanis, E., 2018. Debt consolidation: Aggregate and distributional implications. Unpublished, PhD Thesis, Department of Economics, Athens University of Economics and Business.
- [54] Sakkas S. and Varthalitis P. 2018. The (intertemporal) equity-efficiency trade-off of fiscal consolidation. MPRA Paper 90983, University Library of Munich Germany.
- [55] Schmitt-Grohé S., and M. Uribe. 2007. Optimal, Simple and Implementable Monetary and Fiscal Rules. *Journal of Monetary Economics*, 54 (6): 1702-25.
- [56] Schmitt-Grohé S., and M. Uribe. 2004. Optimal Operational Monetary Policy in the Christian-Eichenbaum-Evans Model of the U.S. Business Cycle. NBER Working paper No. 10724.
- [57] Schwarzmüller, T., and M. Wolters. 2014. The Macroeconomic Effects of Fiscal Consolidation in Dynamic General Equilibrium. *Kiel Working Papers*, 1963
- [58] Trabandt, M., and H. Uhlig. 2011. The Laffer curve revisited. Journal of Monetary Economics, 58(4), 305-327.

A First order conditions

A.1 *Rich* households

Each *Rich* household in any given period t, chooses $c_{R,t}$, $e_{R,t}$, $l_{R,t}$, $k_{R,t+1}$, $b_{R,t+1}$, $h_{R,t+1}$ to maximize its lifetime utility subject to the constraints (4) (in which we incorporate (6) and (7)) and (5) taking factor prices and policy as given. Defining as $\lambda_{R,t}$ and $\psi_{R,t}$ the Lagrange multipliers associated with (4) and (5) respectively. The first order conditions are:

$$\lambda_{R,t} = \frac{\mu_1}{\left(1 + \tau_t^c\right) \left(c_{R,t+1} + \xi g_{t+1}^c\right)}$$
(18)

$$\frac{\mu_2}{1 - e_{R,t} - l_{R,t}} = \psi_{R,t} x_R B_R \theta \left(e_{R,t} \right)^{\theta - 1} \left[\left(e_{R,t} \right)^{\theta} \left(g_{R,t}^e \right)^{1 - \theta} \right]^{x_R - 1}$$
(19)

$$\frac{\mu_2}{1 - e_{R,t} - l_{R,t}} = \lambda_{R,t} (1 - \tau_t^l) w_{R,t} h_{R,t}$$
(20)

$$\frac{1}{(1+\tau_t^c)(c_{R,t}+\xi g_t^c)} = \frac{\beta \left[1-\delta^k + (1-\tau_{t+1}^k)r_{t+1}\right]}{\left(1+\tau_{t+1}^c\right)\left(c_{R,t+1}+\xi g_{t+1}^c\right)}$$
(21)

$$\frac{1}{(1+\tau_t^c)(c_{R,t}+\xi g_t^c)} = \frac{\beta \left(1+r_{t+1}^b\right)}{\left(1+\tau_{t+1}^c\right)\left(c_{R,t+1}+\xi g_{t+1}^c\right)}$$
(22)

$$\psi_{R,t} = \beta \lambda_{R,t+1} (1 - \tau_{t+1}^l) w_{R,t+1} l_{R,t+1} + \beta \psi_{R,t+1} (1 - \delta^h)$$
(23)

A.2 Poor households

Each *Poor* household, P, maximizes its lifetime utility in any given period t by choosing $c_{P,t}$, $e_{P,t}$, $l_{P,t}$, $h_{P,t+1}$ subject to the constraints (9) and (10) taking factor prices and policy as given. Defining as $\lambda_{P,t}$ and $\psi_{P,t}$ the Lagrange multipliers associated with (9) and (10) respectively. The first order conditions are:

$$\lambda_{P,t} = \frac{\mu_1}{(1 + \tau_t^c) \left(c_{P,t+1} + \xi g_{t+1}^c \right)} \tag{24}$$

$$\frac{\mu_2}{1 - e_{P,t} - l_{P,t}} = \psi_{P,t} x_P B_P \theta \left(e_{P,t} \right)^{\theta - 1} \left[\left(e_{P,t} \right)^{\theta} \left(g_{P,t}^e \right)^{1 - \theta} \right]^{x_P - 1}$$
(25)

$$\frac{\mu_2}{1 - e_{P,t} - l_{P,t}} = \lambda_{P,t} (1 - \tau_t^l) w_{P,t} h_{P,t}$$
(26)

$$\psi_{P,t} = \beta \lambda_{P,t+1} (1 - \tau_{t+1}^l) w_{P,t+1} l_{P,t+1} + \beta \psi_{P,t+1} (1 - \delta^h)$$
(27)

A.3 Firms

Each firm f maximizes its profits (11) subject to its production function (12) by choosing k_t^f , $l_{R,t}^f$, $l_{P,t}^f$. The first order conditions are:

$$r_{t} = A\alpha \left(1 - m\right) \left(\rho \left(k_{t}^{f}\right)^{v} + (1 - \rho) \left(l_{R,t}^{f}\right)^{v}\right)^{\frac{\sigma}{v} - 1} \rho \left(k_{t}^{f}\right)^{v - 1} \times \left[m \left(l_{P,t}^{f}\right)^{\sigma} + (1 - m) \left(\rho \left(k_{t}^{f}\right)^{v} + (1 - \rho) \left(l_{R,t}^{f}\right)^{v}\right)^{\frac{\sigma}{v}}\right]^{\frac{\alpha}{\sigma} - 1} (k_{t}^{g})^{1 - \alpha}$$
(28)

$$w_{R,t} = A\alpha \left(1 - m\right) \left(\rho \left(k_t^f\right)^v + (1 - \rho) \left(l_{R,t}^f\right)^v\right)^{\frac{\sigma}{v} - 1} (1 - \rho) \left(l_{R,t}^f\right)^{v - 1} \\ \times \left[m \left(l_{P,t}^f\right)^\sigma + (1 - m) \left(\rho \left(k_t^f\right)^v + (1 - \rho) \left(l_{R,t}^f\right)^v\right)^{\frac{\sigma}{v}}\right]^{\frac{\alpha}{\sigma} - 1} (k_t^g)^{1 - \alpha}$$
(29)

$$w_{P,t} = A\alpha m \left(l_{P,t}^f \right)^{\sigma-1} \left[m \left(l_{P,t}^f \right)^{\sigma} + (1-m) \left(\rho \left(k_t^f \right)^v + (1-\rho) \left(l_{R,t}^f \right)^v \right)^{\frac{\sigma}{v}} \right]^{\frac{\alpha}{\sigma}-1} \left(k_t^g \right)^{1-\alpha}$$
(30)

Notice that each firm, f, makes extraordinary profits given by $\pi_t^f = (1 - \alpha) y_t^f$ as in Guo and Lansing (1997).

B Market clearing conditions

Market clearing conditions in the capital market, the dividends market, the labor (*Rich*-skilled and *Poor*-unskilled) market, the government bonds market are respectively (in per capita terms):

$$n^f k_t^f = n_R k_{R,t} \tag{31}$$

$$n^f \pi^f_t = n_R \pi_{R,t} \tag{32}$$

$$n^{f}l_{R,t}^{f} = n_{R}l_{R,t}h_{R,t} (33)$$

$$n^{f}l_{P,t}^{f} = n_{P}l_{P,t}h_{P,t} (34)$$

$$b_t \equiv \frac{B_t}{N} = n_R b_{R,t} \tag{35}$$

The economy's aggregate resource constraint in per capita terms is given by:

$$n_R \left[c_{R,t} + k_{R,t+1} - (1 - \delta^k) k_{R,t} \right] + n_P c_{P,t} + g_t^i + g_t^c + g_t^e = n^f y_t^f$$
(36)

where we express the number of *Rich* and *Poor* households in terms of their population share shares $n_R \equiv \frac{N_R}{N}$, $n_P \equiv \frac{N_P}{N} = 1 - n_R$.

C Full equilibrium system

The full equilibrium system in per capita terms is given in detail by the following 39 equations in 39 unknowns which are $\{y_t^f, c_{R,t}, c_{P,t}, i_{R,t}, d_{R,t}, k_{R,t+1}, h_{R,t+1}, h_{P,t+1}, b_{R,t+1}, e_{R,t}, e_{P,t}, l_{R,t}, d_{R,t}, h_{R,t+1}, h_{R,$ $l_{P,t}, r_t^b, \pi_{R,t}, \lambda_{R,t}, \lambda_{P,t}, \psi_{R,t}, \psi_{P,t}, k_t^g, r_t, w_{R,t}, w_{P,t}, q_t, g_t^c, g_t^i, g_t^e, tr_{R,t}, tr_{P,t}, g_{R,t}^e, g_{P,t}^e, s_t^{g^i}, s_t^{g^c}, s_t^{g^e}, s_t^{tr_P}, s_t^{tr_R}, \tau_t^c, \tau_t^l, \tau_t^k\}$ given the initial condition for the state variables and the feedback policy coefficients:

$$\lambda_{R,t} = \frac{\mu_1}{(1 + \tau_t^c) \left(c_{R,t+1} + \xi g_{t+1}^c \right)} \tag{37}$$

$$\frac{\mu_2}{1 - e_{R,t} - l_{R,t}} = \psi_{R,t} x_R B_R \theta \left(e_{R,t} \right)^{\theta - 1} \left[\left(e_{R,t} \right)^{\theta} \left(g_{R,t}^e \right)^{1 - \theta} \right]^{x_R - 1}$$
(38)

$$\frac{\mu_2}{1 - e_{R,t} - l_{R,t}} = \lambda_{R,t} (1 - \tau_t^l) w_{R,t} h_{R,t}$$
(39)

$$\frac{1}{(1+\tau_t^c)(c_{R,t}+\xi g_t^c)} = \frac{\beta \left[1-\delta^k + (1-\tau_{t+1}^k)r_{t+1}\right]}{(1+\tau_{t+1}^c)(c_{R,t+1}+\xi g_{t+1}^c)}$$
(40)

$$\frac{1}{(1+\tau_t^c)\left(c_{R,t}+\xi g_t^c\right)} = \frac{\beta\left(1+r_{t+1}^b\right)}{\left(1+\tau_{t+1}^c\right)\left(c_{R,t+1}+\xi g_{t+1}^c\right)}$$
(41)

$$k_{R,t+1} = (1 - \delta^k) k_{R,t} + i_{R,t} \tag{42}$$

$$b_{R,t+1} = b_{R,t} + d_{R,t} \tag{43}$$

$$\psi_{R,t} = \beta \lambda_{R,t+1} (1 - \tau_{t+1}^l) w_{R,t+1} l_{R,t+1} + \beta \psi_{R,t+1} (1 - \delta^h)$$
(44)

$$h_{R,t+1} = (1 - \delta^h)h_{R,t} + B_R \left[(e_{R,t})^\theta \left(g_{R,t}^e \right)^{1-\theta} \right]^{x_R}$$
(45)

$$(1+\tau_t^c)c_{R,t} + i_{R,t} + d_{R,t} = (1-\tau_t^k)(r_t k_{R,t} + \pi_{R,t}) + (1-\tau_t^l)w_{R,t}l_{R,t}h_{R,t} + tr_{R,t} + r_t^b b_{R,t}$$
(46)

$$\lambda_{P,t} = \frac{\mu_1}{\left(1 + \tau_t^c\right) \left(c_{P,t+1} + \xi g_{t+1}^c\right)} \tag{47}$$

$$\frac{\mu_2}{1 - e_{P,t} - l_{P,t}} = \psi_{P,t} x_P B_P \theta \left(e_{P,t} \right)^{\theta - 1} \left[\left(e_{P,t} \right)^{\theta} \left(g_{P,t}^e \right)^{1 - \theta} \right]^{x_P - 1}$$
(48)

$$\frac{\mu_2}{1 - e_{P,t} - l_{P,t}} = \lambda_{P,t} (1 - \tau_t^l) w_{P,t} h_{P,t}$$
(49)

$$\psi_{P,t} = \beta \lambda_{P,t+1} (1 - \tau_{t+1}^l) w_{P,t+1} l_{P,t+1} + \beta \psi_{P,t+1} (1 - \delta^h)$$
(50)

$$h_{P,t+1} = (1 - \delta^h)h_{P,t} + B_P \left[(e_{P,t})^{\theta} \left(g_{P,t}^e \right)^{1-\theta} \right]^{x_P}$$
(51)

$$n_R \left[c_{R,t} + k_{R,t+1} - (1 - \delta^k) k_{R,t} \right] + n_P c_{P,t} + g_t^e + g_t^i + g_t^c = n^f y_t^f$$
(52)

$$(s_t^{g^c} + s_t^{g^e} + s_t^{g^i} + s_t^{tr_R} + s_t^{tr_P})n^f y_t^f + (1 + r_t^b)n_R b_{R,t} = n_R b_{R,t+1}$$

$$\tau_t^c (n_R c_{R,t} + n_P c_{P,t}) + \tau_t^k n_R (r_t k_{R,t} + \pi_{R,t}) + \tau_t^l (n_P w_{P,t} l_{P,t} h_{P,t} + n_R w_{R,t} l_{R,t} h_{R,t})$$
(53)

$$k_{t+1}^g = (1 - \delta^g) \, k_t^g + g_t^i \tag{54}$$

$$n^{f}y_{t}^{f} = A\left[m\left(n_{P}l_{P,t}h_{P,t}\right)^{\sigma} + (1-m)\left(\rho\left(n_{R}k_{R,t}\right)^{v} + (1-\rho)\left(n_{R}l_{R,t}h_{R,t}\right)^{v}\right)^{\frac{\sigma}{v}}\right]^{\frac{\alpha}{\sigma}}\left(k_{t}^{g}\right)^{1-\alpha}$$
(55)

$$r_{t} = A\alpha \left(1 - m\right) \left(\rho \left(n_{R}k_{R,t}\right)^{v} + (1 - \rho) \left(n_{R}l_{R,t}h_{R,t}\right)^{v}\right)^{\frac{\sigma}{v} - 1} \rho \left(n_{R}k_{R,t}\right)^{v - 1} \\ \times \left[m \left(n_{P}l_{P,t}h_{P,t}\right)^{\sigma} + (1 - m) \left(\rho \left(n_{R}k_{R,t}\right)^{v} + (1 - \rho) \left(n_{R}l_{R,t}h_{R,t}\right)^{v}\right)^{\frac{\sigma}{v}}\right]^{\frac{\sigma}{\sigma} - 1} \left(k_{t}^{g}\right)^{1 - \alpha}$$
(56)

$$w_{R,t} = A\alpha \left(1 - m\right) \left(\rho \left(n_R k_{R,t}\right)^v + (1 - \rho) \left(n_R l_{R,t} h_{R,t}\right)^v\right)^{\frac{\sigma}{v} - 1} \left(1 - \rho\right) \left(n_R l_{R,t} h_{R,t}\right)^{v - 1} \qquad (57)$$
$$\times \left[m \left(n_P l_{P,t} h_{P,t}\right)^{\sigma} + (1 - m) \left(\rho \left(n_R k_{R,t}\right)^v + (1 - \rho) \left(n_R l_{R,t} h_{R,t}\right)^v\right)^{\frac{\sigma}{v}}\right]^{\frac{\alpha}{\sigma} - 1} \left(k_t^g\right)^{1 - \alpha}$$

$$w_{P,t} = A\alpha m \left(n_P l_{P,t} h_{P,t} \right)^{\sigma-1} \left[m \left(n_P l_{P,t} h_{P,t} \right)^{\sigma} + (1-m) \left(\rho \left(n_R k_{R,t} \right)^v + (1-\rho) \left(n_R l_{R,t} h_{R,t} \right)^v \right)^{\frac{\sigma}{v}} \right]^{\frac{\alpha}{\sigma}-1} \left(k_t^g \right)^{1-\alpha}$$
(58)
$$\pi_{R,t} = (1-\alpha) y_t^f$$
(59)

The fiscal feedback policy rules are:

$$s_t^{g^c} - s^{g^c} = \gamma_q^{g^c} \left(q_{t-1} - q \right) \tag{60}$$

$$s_t^{g^i} - s^{g^i} = \gamma_q^{g^i} \left(q_{t-1} - q \right) \tag{61}$$

$$s_t^{tr_P} - s^{tr_P} = \gamma_q^{tr_P} \left(q_{t-1} - q \right)$$
(62)

$$s_t^{tr_R} - s^{tr_R} = \gamma_q^{tr_R} \left(q_{t-1} - q \right)$$
(63)

$$s_t^{g^e} - s^{g^e} = \gamma_q^{g^e} \left(q_{t-1} - q \right) \tag{64}$$

$$\tau_t^c - \tau^c = \gamma_q^c \left(q_{t-1} - q \right) \tag{65}$$

$$\tau^l - \tau^l = \gamma_q^l \left(q_{t-1} - q \right) \tag{66}$$

$$\tau_t^k - \tau^k = \gamma_q^k \left(q_{t-1} - q \right) \tag{67}$$

where

$$q_{t-1} \equiv \frac{n_R b_{R,t}}{n^f y_{t-1}^f}$$
(68)

$$s_t^{g^i} \equiv \frac{g_t^i}{n^f y_t^f} \tag{69}$$

$$s_t^{g^e} \equiv \frac{g_t^e}{n^f y_t^f} \tag{70}$$

$$s_t^{g^c} \equiv \frac{g_t^c}{n^f y_t^f} \tag{71}$$

$$s_t^{tr_P} \equiv \frac{n_P tr_{P,t}}{n^f y_t^f} \tag{72}$$

$$s_t^{tr_R} \equiv \frac{n_R tr_{R,t}}{n^f y_t^f} \tag{73}$$

$$g_{R,t}^e \equiv \omega g_t^e \tag{74}$$

$$g_{P,t}^e \equiv (1-\omega)g_t^e \tag{75}$$

D Allowing access to capital/financial markets for *Poor* households

In this Appendix we relax the assumption that *Poor* households do not have access to capital/financial markets. To do this, we allow both *Rich* and *Poor* households to accumulate physical capital and borrow or save in government bonds. As is known, under perfect capital markets and with common discount factors, the allocation of the aggregate stock of capital and bonds to different types of individual investors cannot be pinned down by the equilibrium conditions. To overcome this problem we assume that both types of households face intermediation or transaction costs due to imperfections in capital and financial markets as in e.g. Benigno (2009), Angelopoulos et al. (2014) and Economides et al. (2014). Transaction costs guarantee a well-defined long run solution and induce stationarity during the transition. In addition, they allow us to maintain household heterogeneity in asset holdings (see below). Here we only present the new equations and variables associated with the maximization problem of *Poor* households, and any equations that have changed relative to the model in the main text.

Now, each *Poor* household, P, maximizes its expected discounted lifetime utility subject to the following within period budget constraint which substitutes equation (9) in Section 2.4:

$$(1 + \tau_t^c) c_{P,t} + i_{P,t} + d_{P,t} = (1 - \tau_t^k) r_t k_{P,t} + (1 - \tau_t^l) w_{P,t} l_{P,t} h_{P,t} + t r_{P,t} + r_t^b b_{P,t}$$
(76)

where $i_{P,t}$ is private investment in physical capital of *Poor* households, $d_{P,t}$, is savings in the form of government bonds of *Poor* households, $k_{P,t}$, is the beginning of period physical capital owned by *Poor* households and, $b_{P,t}$, is the beginning of period government bonds owned by *Poor* households. The laws of motion of physical capital and government bonds of each *Poor* household are:

$$k_{P,t+1} = (1 - \delta^k)k_{P,t} + i_{P,t} - \frac{\chi_P}{2} \left(\frac{k_{P,t}}{n^f y_t}\right)^2$$
(77)

$$b_{P,t+1} = b_{P,t} + d_{P,t} - \frac{\phi_P}{2} \left(\frac{b_{P,t}}{n^f y_t}\right)^2$$
(78)

the quadratic terms as in e.g. Benigno (2009), $\frac{\chi_P}{2} \left(\frac{k_{P,t}}{n^f y_t}\right)^2$ and $\frac{\phi_P}{2} \left(\frac{b_{P,t}}{n^f y_t}\right)^2$, capture transaction costs while the parameters $\chi_P, \phi_P \ge 0$ measure the size of these transactions costs. *Poor*

households choose k_P, b_P which yield two new first order conditions:

$$\frac{1}{\left(1+\tau_t^c\right)\left(c_{P,t}+\xi g_t^c\right)} = \frac{\beta \left[1-\delta^k + \left(1-\tau_{t+1}^k\right)r_{t+1} - \chi_P \frac{k_{P,t+1}}{\left(n^f y_{t+1}\right)^2}\right]}{\left(1+\tau_{t+1}^c\right)\left(c_{P,t+1}+\xi g_{t+1}^c\right)}$$
(79)

$$\frac{1}{(1+\tau_t^c)(c_{P,t}+\xi g_t^c)} = \frac{\beta \left[1+r_{t+1}^b - \phi_P \frac{b_{P,t+1}}{\left(n^f y_{t+1}\right)^2}\right]}{\left(1+\tau_{t+1}^c\right)\left(c_{P,t+1}+\xi g_{t+1}^c\right)}$$
(80)

Similarly, the problem of *Rich* households is modified as follows. *Rich* household face similar quadratic transactions costs and thus the laws of motion of physical capital and government bonds for each *Rich* household changes from equations (6) and (7) to:

$$k_{R,t+1} = (1 - \delta^k)k_{R,t} + i_{R,t} - \frac{\chi_R}{2} \left(\frac{k_{R,t}}{n^f y_t}\right)^2$$
(81)

$$b_{R,t+1} = b_{R,t} + d_{R,t} - \frac{\phi_R}{2} \left(\frac{b_{R,t}}{n^f y_t}\right)^2$$
(82)

where the quadratic terms $\frac{\chi_R}{2} \left(\frac{k_{R,t}}{n^f y_t}\right)^2$ and $\frac{\phi_R}{2} \left(\frac{b_{R,t}}{n^f y_t}\right)^2$ capture transaction costs and the parameter $\chi_R, \phi_R \ge 0$ measure the size of these transactions costs.

The first order conditions of *Rich* households with respect to k_R, b_R , change from equations (19) and (21) in Appendix A.1 to:

$$\frac{1}{(1+\tau_t^c)\left(c_{R,t}+\xi g_t^c\right)} = \frac{\beta \left[1-\delta^k + \left(1-\tau_{t+1}^k\right)r_{t+1} - \chi_P \frac{k_{R,t+1}}{\left(n^f y_{t+1}\right)^2}\right]}{\left(1+\tau_{t+1}^c\right)\left(c_{R,t+1}+\xi g_{t+1}^c\right)}$$
(83)

$$\frac{1}{(1+\tau_t^c)(c_{R,t}+\xi g_t^c)} = \frac{\beta \left[1+r_{t+1}^b - \phi_R \frac{b_{R,t+1}}{\left(n^f y_{t+1}\right)^2}\right]}{\left(1+\tau_{t+1}^c\right)\left(c_{R,t+1}+\xi g_{t+1}^c\right)}$$
(84)

For comparison with the benchmark economy, in the numerical solution of section 7.2 we set $\chi_R < \chi_P$ and $\phi_R < \phi_P$ such that the total wealth of *Rich* households is higher than that of *Poor* households. We report that our main results do not depend on the exact values of the transaction costs parameters.

Finally, the resource constraint of the economy changes from equation (36) in Appendix C to:

$$n_{R} \left[c_{R,t} + k_{R,t+1} - (1 - \delta^{k}) k_{R,t} + \frac{\chi_{R}}{2} \left(\frac{k_{R}}{n^{f} y_{t}} \right)^{2} + \frac{\phi_{R}}{2} \left(\frac{b_{R}}{n^{f} y_{t}}^{2} \right) \right] + n_{P} \left[c_{P,t} + k_{P,t+1} - (1 - \delta^{k}) k_{P,t} + \frac{\chi_{P}}{2} \left(\frac{k_{P}}{n^{f} y_{t}} \right)^{2} + \frac{\phi_{P}}{2} \left(\frac{b_{P}}{n^{f} y_{t}} \right)^{2} \right] + g_{t}^{e} + g_{t}^{i} + g_{t}^{c} = n^{f} y_{t}$$

$$(85)$$

while the government budget constraint changes from equation (53) in Appendix C to :

$$(s_t^{g^c} + s_t^{g^e} + s_t^{g^i} + s_t^{tr_R} + s_t^{tr_P})n^f y_t^f + (1 + r_t^b)(n_R b_{R,t} + n_P b_{P,t}) = n_R b_{R,t+1} + n_P b_{P,t+1}$$

$$\tau_t^c (n_R c_{R,t} + n_P c_{P,t}) + \tau_t^k n_R (r_t k_{R,t} + \pi_{R,t} + n_P k_{P,t}) + \tau_t^l (n_P w_{P,t} l_{P,t} h_{P,t} + n_R w_{R,t} l_{R,t} h_{R,t})$$
(86)

The new market clearing conditions for physical capital and bonds are $n^f k_t^f = n_R k_{R,t} + n_P k_{P,t}$ and $b_t = n_R b_{R,t} + n_P b_{P,t}$ which substitute equations (31) and (35) in Appendix B respectively.

To sum up, in the DCE system presented in Appendix C we add four new endogenous variables, $k_{P,t}$, $d_{P,t}$, $b_{P,t}$ and $i_{P,t}$ and four new equations (77), (78), (79) and (80). In addition, the equations (81), (82), (83), (84), (85), (86) substitute equations (42), (43), (40), (41), (52), (53) respectively.

E Tables and Figures

Fiscal				
Instr	$w_R l_R h_R$	$w_P l_P h_P$	rk_R	$r^b b_R$
<u> </u>				
s^{g^c}	0.0071	0.0039	0.0076	-0.3449
s^{g^i}	0.0134	0.0088	0.0122	-0.3394
s^{tr_P}	0.0012	-0.0041	0.0020	-0.3509
s^{g^e}	0.0281	0.0232	0.0289	-0.3240
$ au^c$	0.0081	0.0050	0.0086	-0.3439
$ au^l$	0.0159	0.0138	0.0162	-0.3360
$ au^k$	0.0238	0.0108	0.0116	-0.3343

Table E.1: Steady state labor and capital incomes (as % deviations from status quo)

Figure E.1: Dynamic path of other endogenous variables under the optimized fiscal policy

